REVIEW



Cultural evolution: A review of theoretical challenges

Ryan Nichols^{1,2} [6], Mathieu Charbonneau³ [6], Azita Chellappoo⁴ [6], Taylor Davis⁵ [6], Miriam Haidle⁶ [6], Erik O. Kimbrough⁷ [6], Henrike Moll⁸ [6], Richard Moore⁹ [6], Thom Scott-Phillips¹⁰ [6], Benjamin Grant Purzycki¹¹ [6] and Jose Segovia-Martin^{12,13} [6]

¹Department of Philosophy, CSU Fullerton, Fullerton, CA, USA, ²Center for the Study of Human Nature, CSU Fullerton, Fullerton, CA, USA, ³Africa Institute for Research in Economics and Social Sciences, Université Mohammed VI Polytechnique, Rabat, Morocco, ⁴School of Social Sciences and Global Studies, Open University, Milton Keynes, UK, ⁵Department of Philosophy, Purdue University, West Lafayette, IN, USA, ⁶Research Center 'The Role of Culture in Early Expansions of Humans', Heidelberg Academy of Sciences and Humanities, Heidelberg, Germany, ⁷Smith Institute for Political Economy and Philosophy, Chapman University, Orange, CA, USA, ⁸Department of Psychology, University of Southern California, Los Angeles, CA, USA, ⁹Department of Philosophy, University of Warwick, Coventry, England, UK, ¹⁰Ikerbasque, Basque Foundation for Science, Institute for Logic, Cognition, Language & Information, Bilbao, Spain, ¹¹Benjamin Grant Purzycki, Department of the Study of Religion, Aarhus University, Aarhus, Denmark, ¹²M6 Polytechnic University, Rabat, Morocco and ¹³Complex Systems Institute, Paris Île-de-France, Paris, France Corresponding author: Ryan Nichols; Email: rnichols@fullerton.edu

(Received 14 July 2023; revised 15 January 2024; accepted 16 January 2024)

Abstract

The rapid growth of cultural evolutionary science, its expansion into numerous fields, its use of diverse methods, and several conceptual problems have outpaced corollary developments in theory and philosophy of science. This has led to concern, exemplified in results from a recent survey conducted with members of the Cultural Evolution Society, that the field lacks 'knowledge synthesis', is poorly supported by 'theory', has an ambiguous relation to biological evolution and uses key terms (e.g. 'culture', 'social learning', 'cumulative culture') in ways that hamper operationalization in models, experiments and field studies. Although numerous review papers in the field represent and categorize its empirical findings, the field's theoretical challenges receive less critical attention even though challenges of a theoretical or conceptual nature underlie most of the problems identified by Cultural Evolution Society members. Guided by the heterogeneous 'grand challenges' emergent in this survey, this paper restates those challenges and adopts an organizational style requisite to discussion of them. The paper's goal is to contribute to increasing conceptual clarity and theoretical discernment around the most pressing challenges facing the field of cultural evolutionary science. It will be of most interest to cultural evolutionary scientists, theoreticians, philosophers of science and interdisciplinary researchers.

Keywords: cultural evolution; gene-culture co-evolution; theory; methods; philosophy of science

1. Introduction

The growing attraction of cultural evolutionary science stems from the fact that cultural evolutionary processes operate in ways that can be described as 'descent with modification'. Such processes, like language, exhibit dynamics similar to biological evolutionary processes. This paper serves the research community as a review of theoretical challenges (with occasional suggestions) in cultural evolutionary science across four domains: conceptual challenges; methodological issues, including modelling, experimentation and field studies; topical challenges; and implications and applications. In the remainder of this section, we define terms and restate survey results conducted by the Cultural Evolution Society, results which we use to motivate the rest of the discussion.

© The Author(s), 2024. Published by Cambridge University Press. This is an Open Access article, distributed under the terms of the Creative Commons Attribution-NonCommercial-NoDerivatives licence (http://creativecommons.org/licenses/by-nc-nd/4.0), which permits non-commercial re-use, distribution, and reproduction in any medium, provided that no alterations are made and the original article is properly cited. The written permission of Cambridge University Press must be obtained prior to any commercial use and/or adaptation of the article.

The term 'cultural evolutionary science' (abbreviated 'CES') refers to an interdisciplinary field of inquiry using an array of scientific methods to study culture and the interactions between cultural and biological evolutionary processes. Contents of cultural evolutionary science include observation statements, methodological commitments, presuppositions about varieties of statistical tests, and more. 'Cultural evolutionary theory' ('CET') refers to the subset of cultural evolutionary science composing laws, generalizations and hypotheses, and their derivations and entailments. The events, processes and objects in the world that physically and mentally instantiate the transmission and evolution of culture across persons, populations and generations we refer to simply as 'cultural evolution' or 'cultural evolutionary processes'.

Numerous reviews of CES have been written. Creanza et al. (2017) review CES results broadly relevant for demography and genetics. Legare (2017) reviews adaptations for teaching and imitation and their cultural representation. Mesoudi (2016) makes a broad review of findings in CES. Singh et al. (2021) review CES findings that, per the title, go 'beyond social learning'. Smith (2020) reviews theory and findings about cultural group selection. Tamariz (2017) focuses on cultural language evolution. Whiten (2019) reviews experimental findings pertaining to culture in non-animals. Smolla et al. (2021) focus on underappreciated features of cultural evolution. However, these papers have not attempted to diagnose theoretical problems in CES whose varied instantiations animate the discussion below. This is no criticism since the aims of these other review papers are mainly to summarize recent empirical results (usually within one sub-area of CES) and identify potential hypotheses for further testing. The present contribution is instead a critical review of the needs and challenges facing the body of theory accompanying CES, which we refer to as cultural evolutionary theory.

An efficient means for researchers concerned with CET to contribute to improvements in the field is to take account of and respond to the most pressing stated challenges facing the field. As a result, the present critical review of theoretical needs takes its cue from results of a recent survey of the membership of the Cultural Evolution Society about the field's 'grand challenges' (Brewer et al., 2017). Published results of this survey detail two varied sets of challenges. The first set was drawn from semantic analysis of topics directly from survey results. (Subsequent references to 'survey results' are to this paper.) Its most pressing challenge, revealed to be about twice as important as the next topic, was with 'knowledge synthesis'. The authors state that this topic relates 'primarily to issues of theoretical integration and speaks to the idea that while many behavioural scientists and humanities scholars see culture as a defining feature of humankind, different subfields rarely read each other's work or build interdisciplinary research programmes to explore how human cultures differ from those of other animals' (Brewer et al., 2017: 1). Additional topics of appreciable concern in the first set of results were topics labelled 'culture definition', 'theory', 'shared language' across the disciplines, 'pro-sociality' and 'cultural transmission'.

The second set of 'grand challenges' were produced by co-authors of the report following their reflection on the first set of results and the clustered relations amongst those results. In the second set are:

- (C1) understanding 'social adaptation' and what drives it;
- (C2) understanding 'the role of cultural evolution in the context of organic evolution' in light of the fact that 'culture is strongly influenced by biology';
- (C3) overcoming problems involved in 'modelling culture as a complex adaptive system';
- (C4) 'identifying processes of transmission and accumulation of cultural traits' to reach 'consensus on the mechanisms underlying' mathematical models of cultural evolution;
- (C5) 'integrating methods, data, and results across disciplines', which is reported to necessitate 'shared conceptual references and vocabulary';
- (C6) creating 'organizational and funding structures';
- (C7) 'identifying cultural evolutionary processes that address significant social, economic, and political problems' including 'climate change'; and
- (C8) 'educating policymakers' and developing a public understanding of results of cultural evolutionary research (Brewer et al., 2017: 3).

One problem is symptomatic of several of these challenges.

Often CES research teams identify a concept representing some coarse-grained phenomenon; the concept is operationalized for use in modelling, laboratory experiments and/or field studies; corresponding empirical research is conducted and intriguing results are found. Often this occurs without a definition or formal analysis of (i.e. without providing a set of necessary and sufficient conditions to govern) the concept. (See Lewens 2020; we thank Tim Lewens for crystallizing this problem for us.) Tokens of this type of what we will call the 'general problem' not infrequently issue in enigmatic results, methodological challenges, misinterpretation or miscommunication. The presence of this general problem is unsurprising to those who believe that even the term 'cultural evolution' is 'confused' because it is 'being used to indicate both a phenomenon – culture changing through time – and an approach to study it' (Micheletti et al., 2022: 667). Echoes of this general problem lurk behind problems of 'knowledge synthesis' and 'theory', and concerns with definitions of 'culture', 'cumulative culture' and 'social learning', as well as challenges (C1)–(C5).

The primary goal of this paper is to contribute to improving future research in CES by increasing the conceptual clarity and theoretical discernment around several of the most pressing challenges facing CES as revealed by these survey results. These challenges range widely and frequently overlap, which informs the paper's eclectic organization. Section 2 opens with some case studies to establish the utility of a focus on theoretical issues. Section 3 reviews germane developments in the philosophy of science. Section 4 focuses attention on methods challenges in CES, especially on issues of integration as found in (C5). Section 5 discusses two topical issues, culture in human and non-human animals and the nature of 'selection' in linguistics. Taking a cue from (C7) and (C8), Section 6 discusses challenges facing application of CES to policy, with attention to climate change.

2. Case studies in the value of theory and conceptual analysis for CES

To motivate the utility of critical discussion of the field's theoretical challenges, we lead with three abbreviated case studies. Each represents one area in which survey results have pointed to a 'grand challenge'. Each case study anticipates a subsequent section of the paper.

2.1. Conceptual challenges

Conceptual challenges are typically problems of definitional scope. The importance of establishing clear definitions of terms is obvious from survey results that lament the field's lack of 'shared conceptual references and vocabulary' and its need to find 'common language' (Brewer et al., 2017: 3). Meanings of 'copying', 'imitation' and 'social learning' are used equivocally across CES, impeding consistent operationalization in experiments. An influential formal model of cultural transmission purports to show that 'environmental learning' does not, but 'social learning' does, produce the 'empirical 'S-shaped' cumulative adoption curves that dominate the diffusion of innovations literature' (Henrich, 2001: 992). (The term 'S-shaped curve' refers to a sigmoid curve plotting data about the adoption of innovations characterized by, first, lengthy, laboured growth of early adopters followed by a quick increase in adoption, creating a high slope, and concluded by a slowly increasing phase with low slope and many adopters.) Tim Lewens argues (2015: 114-118) that this model treats its concepts of 'environmental learning' and 'social learning' as mutually exclusive. Henrich's definition of 'environmental learning' reads: 'individuals acquire and evaluate payoff-relevant information about alternative behavioural options by action and interaction in their local social, economic and ecological environments' (Henrich, 2001: 992). As defined, the meaning of 'environmental learning' overlaps with the meaning of 'social learning'. That is, the two are not independently defined. This entails that 'environmental learning' could also be modelled to yield S-shaped curves. This appears inconsistent with Henrich's conclusions from the model.

If the definitions of 'social learning' and 'environmental learning' are not retooled, subsequent modelling and experimentation using these definitions are unlikely to overcome challenge (C3).

4 Ryan Nichols et al.

This conceptual misstep appears to inform pointed empirical criticisms of the model's explanation (Henrich, 2004) of the contraction of the Tasmanian toolkit (Read, 2011; Vaesen et al., 2016a, b; Henrich et al., 2016). This preliminarily demonstrates the value of critical philosophical attention on definitions in CES; see Section 3 below.

2.2. Methods

'Cumulative cultural evolution', a topic of significant concern in survey results pertaining to the dynamics of cultural systems (Brewer et al., 2017: 2), refers to a process in which traditions are modified and improved over time with a performance capacity that is beyond what an individual could generate on her own. This marks a *necessary condition* for a process to represent cumulative culture, a necessary condition widely repeated amongst experimentalists. For example, in their experimental attempt to understand what they call 'cumulative culture', Derex et al. (2013) repeat this condition in their first paragraph by writing that cumulative culture is that which 'no individual could have invented alone' (p. 389). This necessary condition on cumulative culture was first articulated by Boyd and Richerson (1996: 80), who wrote that for a process to be properly cumulative, it must lead human cultures to 'accumulate changes over many generations, resulting in culturally transmitted behaviors that no single human individual could invent on his own' (Boyd & Richerson, 1996: 80; our italics). Thus, to test whether a cumulative effect has been obtained, an experiment must at minimum compare how a solitary individual and a diffusion chain fare on the same task using equivalent time resources (Miton & Charbonneau, 2018: 3).

Study of cumulative culture is done through a variety of methods including modelling, field studies, experiments and more, but this issues in a challenge of creating necessary and sufficient conditions (see (C5) at Brewer et al., 2017: 3) that are clear, consistent and integrated across methods. According to Miton and Charbonneau (2018), this is yet to happen in the study of cumulative culture. Specifically, a crucial difference emerged between theory-based analysis of the cumulative culture concept, in which the necessary condition was identified, and its operationalization in experiments and models. The necessary condition stated above is infrequently operationalized in experiments. And when it is (as in Derex et al., 2014; Mesoudi, 2011), it risks misleading readers. Suppose a solitary individual and a group in a diffusion chain are each given one hour to complete the same task, and suppose that the group completes the task faster. Is the necessary condition met by this operationalization? No. Miton and Charbonneau (2018: 3) argue that this is 'insufficient to warrant the conclusion that observed differences in performance are the results of a genuinely cumulative process' because these control conditions cannot distinguish between performance capacity (evidence of cumulative culture) and its confound of improvement speed. This is because participants might have their individual skill development accelerated by learning in a social context (improvement speed), and complete a puzzle, even though social collaboration across generations is unnecessary (performance capacity) for completion of the puzzle (Miton & Charbonneau, 2018: 3). Section 4 includes discussion of challenges (C3), (C4) and (C5) pertaining to three common methods in CES research, modelling, experimentation and field studies.

2.3. Topical challenges

Sub-disciplines within CES face messy challenges when researchers port coarse-grained focal concepts from one subfield to another. In the Cultural Evolution Society survey (Brewer et al., 2017: 1–2), the challenge of synthesizing knowledge across diverse research areas was rated as the field's most pressing concern. Consider the status of a prominent CES explanation of 'cooperation', which says that cooperation within groups emerged because of psycho-social consequences of belief in punitive gods (Johnson, 2005; Norenzayan, 2013). This causal claim was supported by diverse studies ranging from anthropological databases that identified statistical relations between a culture's adoption of a moralistic high god and markers of cooperation (Roes & Raymond, 2003) to economic games

experiments that showed priming with religious concepts raised offers to strangers (Shariff & Norenzayan, 2007), and from cognitive science of religion research on minimally counterintuitive ideas (Norenzayan et al., 2006, although see Purzycki & Willard, 2016) to psychological studies of agent-based teleological thinking (Kelemen, 1999).

Yet CES-inspired study of religion and cooperation can succumb to the aforementioned challenges. As part of a wider problem in the social sciences, religious priming studies have suffered from multiple failures to replicate (e.g. Gomes & McCullough, 2015; Miyatake & Higuchi, 2017). While cross-cultural experimental evidence bolsters the supernatural punishment–cooperation relationship (Lang et al., 2019), the purported causal relation between supernatural punishment and social complexity (an index of broader cooperation) remains unclear (Watts et al., 2015) and more basic conceptual concerns about presumed psychological mechanisms linger (Schloss & Murray, 2011). Echoing our 'general problem', critics argue that the concept of 'cooperation' at work in models and experiments about effects of belief in supernatural punishment on cooperation often equivocates because within-group parochiality is confused with cooperation between non-kin (Galen, 2016).

Critics might argue that research on this topic is consistent with scientific progress owing to its self-correcting nature, and as a result, the community does not require theory-driven re-evaluation of the subfield but more detail-oriented diligence. While we could be incorrect, it seems that focus on prosaic aspects of CES research can camouflage hurdles still thwarting progress. Most hypotheses tested in this subfield were not logical derivatives from a principled theory in part because the concept of a 'high god' lacks adequate formality and consistency across operationalizations. Some advocates define the term 'high gods' as 'gods who cared about cooperative – and harmony – enhancing behavior' (Shariff, Norenzayan, & Henrich, 2010: 124); others defined 'high gods' as deities specifically responsible for creating reality, and who may or may not be 'specifically supportive of human morality' (Swanson, 1960). Confusions of this type put researchers' ability to falsify causal hypotheses about religious beliefs and cooperation at risk since, if a correlation was not discovered in a dataset, the former circular definition warrants the inference that the dataset represented a 'low' not a 'high' god (Nichols et al. (2020: 4), and see Table 1 in that paper for similar logical problems in this subfield; see Lightner et al. (2023), on the empirical implications of these conflations). Below, Section 5 analyses theoretical difficulties as found in the topical areas of non-human culture and language.

2.4. Implications and applications

There are many appreciable consequences of CES findings for public policy, social change, well-being and research ethics. Aside from a few exceptions (e.g. Wilson, 2011) these implications have not been thoroughly studied. Section 5 seeks to open this area with discussion of one application (about norms) and one implication (about human uniqueness).

These case studies provide ample justification for the value of theoretical and philosophical attention given to major challenges to CES. To improve the conceptual clarity and theoretical discernment in ongoing CES research, and because it may be unfamiliar to readers, we next encapsulate extant philosophy of science research devoted to challenges voiced in the survey results above.

3. Developments in the philosophy of science for cultural evolution

The contemporary field of cultural evolutionary science has its formal origins in the work of Luigi Luca Cavalli-Sforza and Marcus Feldman (1973) and Marcus Feldman and Cavalli-Sforza (1976). Borrowing modelling strategies from population genetics and epidemiology, they identified analogies between biological and cultural evolution and transmission, exploring culture as a distinct form of inheritance. Subsequent generations of researchers have fleshed out this modelling approach, developing it further in multiple directions and identifying new explanatory projects – opportunistically borrowing tools from evolutionary biology, quantitative and cultural sociology, anthropology and cognitive psychology, among others. As a result, even the foundational status of formal mathematical

models has not led to anything like a single, paradigmatic approach in CET. Instead, the field's development reflects its eclectic opportunism, leaving a vibrant patchwork of loosely related approaches vulnerable to the aforementioned challenges, as well as to explicitly theoretical problems not mentioned in survey results but raised in recent philosophy of science.

In the first wave of theoretical and philosophy of science work on cultural evolution, applications to culture of the Modern Synthesis in biology were developed (see Griesemer, 1988; Hull, 1988; Wimsatt, 1999). A second wave emerged at the end of the twentieth century. While continuing to involve reapplication of tools from biology to culture (e.g. see Skyrms, 2014 on evolutionary game theory), in this period additional effort was placed on analysis of competing theoretical frameworks for CET (see Sterelny, 2017), understanding conditions for culture in non-human animals (e.g. Ramsey et al., 2007; Sterelny, 2009) and evaluating the special character of cultural evolutionary explanations (e.g. Godfrey-Smith, 2012). Currently, a third wave of theoretical work in CET is starting to swell (see e.g. Lewens, 2015) in response to the emergence of problems for CES without precursors in biology. Theoreticians and philosophers of science are now elaborating methods specific to CET, including modelling, experimentation and field studies, observing operational challenges in each that follow from the use of concepts like 'culture' and 'cumulative culture' (Buskell, 2022; Sterelny, 2021; Vaesen & Houkes, 2021), 'prestige bias' (Chellappoo, 2021) and 'cultural attractors' (Driscoll, 2011; Buskell, 2017). We contextualize then highlight three sets of philosophical problems that, in effect, we seek to add to the list of grand challenges facing the field.

First, CES sources generally deny the claim, originally by Dawkins (1989) about 'memes', that cultural transmission is replicative (Lewens & Buskell, 2023; see Henrich et al., 2008 & Sperber, 2000; Dennett, 2017 is an exception). Nonetheless, many CES researchers appeal to 'high-fidelity copying' to explain complexity and diversity in human culture. Befitting the early growth of a new scientific endeavour, we generally welcome explanatory pluralisms (Charbonneau, 2020), yet in this case it represents a theoretical problem for the field. In biology, since genotype to phenotype mapping exists, and since transmission occurs from genotype to genotype, a principled 'grain' of description exists to guide the scientific study of transmission. This is absent in culture. The following example illustrates the 'grain' problem, which can be considered an example of the general problem noted above. 'Suppose Andrew sings a line from a song, and Tim attempts to copy him: does "high fidelity" learning require Tim to reproduce the same melody, must he sing in the same key, must he sing with the same words, and the same emotional emphases?' (Lewens & Buskell, 2023: section 6.2). If Tim sings with different words, in a dissimilar key, or with new accompaniment, under what conditions is it appropriate to reason that he has 'copied' Andrew? The specific means by which CES researchers measure fidelity or copying often lead to incompatible and incommensurable results, challenging the idea that fidelity would be a general, natural property of human cultural learning, let alone a causal one (Charbonneau & Bourrat, 2021).

Second, the advent of CES practised in broad spatiotemporal frames includes projects that consolidate historical and cross-cultural data in large databases (Slingerland et al., 2020), efforts in evolutionary archaeology and cognitive archaeology (Killin & Pain, 2023), and in cultural macroevolution more generally (Gray & Watts, 2017). While surveyed members of the Cultural Evolution Society regard the definition of 'culture' as a vexed but important problem (see Driscoll, 2017; Ramsey & De Block, 2015), a corollary often goes unnoticed, viz. that individuating cultures is distinct from the problem of defining 'culture' *simpliciter*. So what are 'cultures' or 'cultural groups' and how should they be individuated? Are cultures natural kinds? The latter metaphysical question is not idle: many working in political philosophy have long alleged that talking about 'cultural groups' is erroneously essentialist (Patten, 2014). If we wish to avoid essentialist characterizations of specific 'cultures', then how can cultural groups be taxonomized in modelling or experimentation? Related, how do various elements of cultures, for instance, texts and material technologies, relate to one another (Buskell et al., 2019)?

Third is a set of issues emerging from the role of values in science, including epistemic values, absent from survey results. 'Epistemic values' is a term referring to normative traits of theories or methods themselves. Different methods and theories implicitly prioritize some values or purposes

over others. For example, experimental collider physics prioritizes certainty over generalizability owing to its endorsement of a 5σ criterion of statistical significance. Cultural evolutionary science tends to emphasize the epistemic value of generalizability in part because of its origins in modelling. Generalizability can motivate the development of thin, abstract and largely ahistorical theories (Fracchia & Lewontin, 1999). Epistemic values of CES in turn support characterizations of human cognition that are automatic and 'kinetic' (Lewens, 2015), a term used to refer to the population modelling approach to cultural evolution as akin to treating individuals as undifferentiated molecules familiar from the kinetic theory of gasses, and thus, far easier to formally model. Extant, 'thin' characterizations of cognition can pose a special problem in CES because they may misrepresent actual enculturation and transmission of cognitive contents, and also because they may feed narratives about culture and cultural change according to which individuals, perhaps especially in non-Western cultures (Wolf, 2005; Chellappoo, 2022a), lack agency. Generally, what are the proper roles of personal and sub-personal processes in formal explanations of cultural transmission? This question gets to the heart of differences between the CES frameworks of cultural attractor theory and of population modelling (see Buskell, 2019). Is it appropriate for 'population-thinking' models to minimize agency by emphasizing roles of copying and imitation while neglecting to model roles of coercion or institutions (Boyd & Richerson, 1985)? Does 'cultural attractor theory' (Sperber, 1996) in contrast attribute to agents too much capacity to modify cultural items while transmitting them? Increased efforts to integrate modelling with subtler accounts of encultured human cognition and transmission, in which they are sometimes rational, effortful and culturally inflected, will probably improve the field.

Finally, current research in philosophy of science points to the promise of bootstrapping CES and CET research into a *philosophy of nature*, i.e. formulating an 'overall picture of the natural world that science [...] seems to be giving us' (Godfrey-Smith, 2001: 284). Important contributions include Kim Sterelny's (2012, 2021) work aimed at outlining the natural history of human beings and their hominin ancestors, and work making sense of the striking increase in technological complexity of hominins (Pain & Brown, 2021). This marks a promising shift that some have described in terms of movement from a constrained form of behavioural innovation to one that is 'open-ended' (Borg et al., 2022; Charbonneau, 2015), perhaps in a way analogous to the open-ended possibilities afforded by genetic recombination (Charbonneau, 2016).

4. Methods: Models, experiments and field studies

In this section, we discuss challenges pertaining to three common methods in CES. Giant strides have been made in the refinement and sophistication of these scientific techniques in recent decades. The body of local theory tethered to a single method is used to warrant choices in modelling (e.g. about framework or scale), in experimentation (e.g. about deriving from theory a hypothesis for testing) and in field studies (e.g. about operationalizing 'social learning' in archaeological contexts). The local theoretical reflection on decision-making about model type, parameterization and more, is rich and growing richer. This type of 'method-specific' theory is distinct from and narrower in scope than the cross-disciplinary theory essential for improving knowledge integration and meeting other needs of the CES research community. For example, even though an agent-based model might be exquisitely motivated on its own terms, there are rarely to be found theoretical tools that provide a principled structure for how the results of such a model are to be validated using data or results from experiments or field studies, even when those methods are used to explore ostensibly similar research questions.

4.1. Models in cultural evolution

As cultural evolution developed into a scientific field, innovative mathematical and computational techniques were devised to represent intricate features of cultural processes (Kandler et al., 2012). Broadly speaking, three important modelling frameworks have helped to improve our theoretical

understanding of evolutionary processes. The first framework is composed of *gene-culture coevolution models*, which assume a continuous, bidirectional interaction between genes and culture (Cavalli-Sforza & Feldman, 1981; Boyd & Richerson, 1985). Next are *cultural neutral models*, which assume that all variants are equally likely to be adopted, and attempt to understand cultural evolution in the absence of selective forces by assuming that changes in cultural variants are due solely to random drift and innovation (Bentley et al., 2004; Blythe, 2012; Reali & Griffiths, 2009). Finally, the *cultural selection* framework assumes that some variants spread more rapidly than others, making evolution a process that is not neutral, but subject to biases (e.g. systematic cognitive and transmission patterns deviating from rationality) or selection pressures (e.g. environmental factors that affect fitness of variants) (Boyd & Richerson, 1985; Hagen & Hammerstein, 2006; Henrich & McElreath, 2007; Vale et al., 2017).

While modelling frameworks make distinctive assumptions about the nature of cultural processes, these frameworks can each be instantiated at either of two common scales. Regarding the scale used to model cultural evolution, researchers in the field have distinguished between two general classes (Van Dyke Parunak et al., 1998; Kandler et al., 2012). The first is composed of *micro-scale models*, which aim to simulate how individual units in a system behave and interact with one another. The best-known type of micro-scale model is the agent-based model, which consists of a system composed of a set of autonomous units called agents, which simulates the decision-making, behaviour and interactions of agents to understand how patterns emerge at the population level. Such models have been used to study segregation dynamics (Schelling, 1971), the emergence of cooperation (Axelrod & Hamilton, 1981), the evolution of language (Centola et al., 2007; Kirby et al., 2014) and network dynamics (Segovia-Martín et al., 2020), and to adjudicate between competing explanations for biocultural phenomena like footbinding (Chowdhury et al., 2022).

The second class is composed of *macro-scale models*, which attempt to capture the overall behaviour of the system. Macro-scale models may consist of ordinary or partial differential equations that contain one or more functions with their derivatives. In this context, the functions represent quantities (e.g. populations A and B), the derivatives define their rates of change and the equation defines the relationship between both variables. These models have been used to study topics such as language competition (Abrams & Strogatz, 2003), political competition (Misra, 2012), the evolution of religiosity (Rowthorn, 2011), the distribution of cultural traits (Manrubia & Zanette, 2002) and innovation (Kandler & Laland, 2009), among other social issues (see Brown, 2007). Other macro-models, such as phylogenetic or statistical inference models, are oriented to encompass the broader evolutionary relationships between organisms and groups using existing experimental, historical and archaeological data (Turchin et al., 2012).

An appreciable challenge facing the use of modelling to draw substantive inferences about actual cultural processes is that varied costs involved in selection of a framework and a scale for analysis often limit the inferential range of the model's results. Micro-scale models allow more control over the properties of individual entities in the system, generally at the expense of higher computational cost and greater complexity of formalization, which limits their reproducibility and generalizability. Macro-scale models, on the other hand, allow greater control over population variables and the evaluation and formalization of the dynamical system, although at the cost of a lack of control over the individual entities of the system. From these two types of models emerged a general theoretical concern over how to integrate their results.

Of very special consideration in models of cultural evolution is the development of techniques to represent multivariate causal interactions between phenomena at both micro- and macro-levels, e.g. the 'ratchet effect' (Tennie et al., 2009), often regarded as indispensable for explaining cumulative cultural evolution. Recent formal developments enabling the combination of micro- and macro-models are allowing modellers to meet this need and, consequently, they will stimulate high-quality research in CES and assist in overcoming challenge (C3). Micro- and macro-models can be combined by building adaptive dynamics models or with multilevel selection models where both events at the individual level generate effects at the group level and events at the group level affect the individual properties of the

system (Mullon & Lehmann, 2017; Fishwick, 1995; McElreath & Henrich, 2007; Waring et al., 2017; Segovia-Martin & Tamariz, 2021). This represents the future of modelling in cultural evolution, but it is not itself without challenges.

Despite numerous advantages, joint micro-macro models are now confronting novel theoretical problems. These include reconciling the role of individual-level behaviours with aggregate-level phenomena; addressing emergent properties that cannot be explained by the actions of individual agents alone; accounting for feedback loops between the micro- and macro-levels; and specifying the content, behavioural influence and population-level effects of 'regulatory traits', i.e. traits governing cultural transmission of other traits (Acerbi et al., 2014). Given their content, progress on these problems would benefit from formal analyses of concepts like *collective action*, *coercion* and *institutions*, which would facilitate operationalization in models.

4.2. Experiments in cultural evolution

Challenge (C3) focuses on the need to overcome several problems facing the effort to model culture as a complex, adaptive system. In their report of survey results, authors wisely distinguish this from a corollary challenge, (C4), which involves the need to develop an evidential bridge between modelling results and research on the psycho-social mechanisms instantiating cultural transmission in the world. Given the ascendant place of modelling in CES, validating models against data from experiments and field studies has remained a persistent challenge. This challenge for CES typically has one of four points of inception: (i) a lack of data, making validation unfeasible; (ii) a conceptual gulf between model and empirical data; (iii) limitations of the model's ability to properly describe new, previously unseen data; and (iv) a lack of parallelism between model mechanisms and actual mechanisms, making models of little use in capturing the causal interactions between real evolutionary processes (see Weisberg, 2007, on assignment, scope and fidelity). Mechanisms for cultural evolution like social learning biases have of course been proposed (Boyd & Richerson, 1985), modelled (Denton et al., 2020) and experimented upon (Efferson et al., 2008). Yet CES nonetheless faces several problems integrating theory and modelling with experimentation often owing to the absence of a set of criteria for a measurable outcome that is commensurable across methods.

Take, for example, research that frames lab- and field-based experiments (see Mesoudi & Whiten, 2008 and Tamariz, 2017 for reviews). Cultural transmission has been a focus of laboratory experiments, many of which examine whether or not some of the postulated learning biases exist. Other studies pit multiple learning biases against each other to assess which are more powerful explanations of cultural retention and transmission. While some experiments focus on biased social learning, including, among others, content- (e.g. Boyer & Ramble, 2001; Broesch et al., 2014) and prestigebiased cultural transmission (e.g. Jiménez & Mesoudi, 2019) in the retention and recall of stimuli, others utilize sequential transmission chains to examine ostensibly unique properties of human cultural evolution such as cumulative evolution and problem-solving (e.g. Caldwell & Millen, 2008; Jiménez & Mesoudi, 2020). In step with the classic experiments of Bartlett (1932), many of these studies follow the logic and structure of 'telephone', a game that exploits distortion of information across chains of multiple learner-teachers. Other experiments stress communication and group coordination through actively and collectively aligning individual behaviours, such as drumming and movement, and examining downstream behaviour (e.g. Tunçgenç & Cohen, 2016; Wood et al., 2018). Aptly, but unsystematically, the range of participants' agency varies significantly across cultural transmission experiments. This also applies to studies of cultural evolution in non-human animals, where a variety of evidence stemming from controlled experiments and anchored in social learning models has been observed (Whiten, 2019).

Much of the behavioural experimental work that falls under the rubric of cultural evolution assesses the magnitude and scope of cooperation that is unique to our species. By measuring individual decisions in allocating money and other resources, experiments in cross-cultural behavioural economics suggest that culture and cultural institutions play a significant role in promoting more cooperative

strategies (Pisor et al., 2020) or, less appreciated in CES, in *reducing* cooperation (cf. Akbari et al., 2019). Studies that deploy the same experimental game but have culturally salient treatment conditions show that when made culturally relevant, salient beliefs and values can alter the course of interpersonal behaviour (Cronk, 2007; Gerkey, 2013; Lightner et al., 2017). In such experiments, the outcome variable is cooperation, not whatever cultural information that is purportedly inducing it, and certainly not the transmission of such cultural information. Yet some paradigms have exploited such experiments, and rather than using beliefs and cultural framings to predict behaviour, some have reversed this structure and examined how game outcomes can predictably shift the content and structure of extant beliefs (Purzycki et al., 2020). Neither of these experimental paradigms directly speaks to transmission, however; one seeks to explain the evolution of cooperation and the other seeks to explain the evolution of beliefs as responses to threats to cooperation.

While the ultimate theory that motivates such studies posits that cultural norms can co-evolve with problems of cooperation, there remains a notable deficit in just how little of this work specifically attends to cultural transmission and the co-variation between cooperation and tradition. Experiments about variability of dynamics in transmission across cultures are infrequently motivated by integration with cross-cultural psychology and so, albeit with exceptions (e.g. Mesoudi et al., 2014, which found significant cross-cultural variation in the adoption rate of success-based social copying), they generally do not motivate or test explicit hypotheses about cultural differences in transmission. Many such studies might assess the cultural mechanisms that promote or reduce cooperation but do nothing in the way of accounting for the diffusion or maintenance of such beliefs. In other words, while we might see co-varying effects of cultural information and behaviour in the lab, and while this might make certain explanations or theories more plausible, it remains unclear what this has to do with the evolution and persistence of the cultural part of the dynamics. Are particular cultural strains prevalent because they promote (or reduce) cooperation? Do the social benefits of cooperation contribute to the prevalence and maintenance of particular cultural mechanisms? Such questions are ripe for serious experimental treatment, and returning to challenge (C4), the results of such experiments are expected to inform representations of subtle, varied causal interactions in future micromacro models.

4.3. Field studies

Field studies fall into various kinds. Ethological field studies concern cultural performances beyond our species. Archaeological and palaeoanthropological excavations provide insight into the deep past and into the emergence of cultural evolution of hominins from australopithecines and different Homo species up to anatomically modern humans. Anthropological field studies, often separated into cultural anthropology and evolutionary anthropology, examine social and material interactions in present-day human groups and societies and their cultural products. Although these three fields differ markedly in the nature and acquisition of their basic data as well as in their theoretical presuppositions, they share important commonalities. Key features of all three approaches are: adoption of long-term data collection methods and analysis; attention to facts about the material, ecological and behavioural contexts of cultural behaviours; and interest in identifying a potential variety of contributing elements. Field studies in CES aim to gain a broad overview of cultural performances and their environmental conditions, and to examine the differences between distinct social and ecological settings in order to increase understanding of the historical and biological circumstances of cultural evolution. Field studies in CES incorporate fuzzy evidence about multi-faceted entanglements between agents, environments and behaviours, captured with a variety of data types ranging from anecdotal reports to behaviour coding to structured research protocols, within which meaningful patterns of relations and development can be detected.

Field studies have played and will continue to play the most prominent role in CES responses to challenge (C8), which encompass the need to convince policymakers of the need to conserve animal cultures. Until the second half of the twentieth century, culture had only been ascribed to humans.

Since the expansion of the definition of culture to make it applicable to non-human animals (McGrew, 1992), and subsequent studies on behavioural traditions in chimpanzees (Whiten et al., 1999), which combined field research from several groups in search of differences of occurrences of behaviour across sites under similar environmental conditions, ethological field studies with cultural evolutionary scope have been on the rise. The variety of animal species examined as well as evidence of performances with basic cultural features have markedly increased. Ethological research has produced studies on the pace of animal innovations, as found in seasonal changes in songs of humpback whales (Schulze et al., 2022), on cultural transmission in the manufacture of tools, as in New Caledonian crows (Hunt & Gray, 2003), and on the adoption of new behavioural elements from resident individuals when they must adapt to new habitats after dispersal, as in male orangutans (Mörchen et al., 2023). The current state of the field suggests that there is a variety of basic cultural performances in a wide array of animal orders from invertebrates to birds and mammals (Whiten, 2021) and that the number of (unspecified) cases of 'culture' is underestimated (Schuppli & Van Schaik, 2019).

In contrast to field studies on animal behaviour and ethnography, which include direct observations of performances, primary data from archaeological field studies provide only indirect insights into past human behaviour through trace material evidence. Performances, and thus also groupspecific patterns and their development, must be inferred from finds via multiple analytical steps before interpretation (Perrault, 2019). The chronological frame of archaeological field studies spans more than 3 million years. Research focused on distinct spans of time within this epochal stretch has enabled data collection from the material record to respond to quite different questions about cultural evolutionary processes. Inspired by behavioural studies on animals, evolutionary archaeological research focusing on an early phase up to 500,000 years ago (Shennan, 2008) is dominated by the distinction of primate and hominin cultural capacities (Tennie et al., 2017), long-lasting technical traditions and convergent evolution in technologies (O'Brien et al., 2018), and the development of simple stages of communication and cooperation (Gärdenfors & Högberg, 2017). For a middle phase between 500,000 and 100,000 years ago, the research focus shifts to the study of evidence for the increasing complexity of technical expressions (Lombard & Haidle, 2012) and the necessity of advanced forms of cultural transmission (Stolarczyk & Schmidt, 2018), group structures and the advent of ritual behaviour (Dapschauskas et al., 2022). In a late phase from 100,000 years on, relations of techno-complexes and social systems as well as the development of group identities (Vanhaeren & d'Errico, 2006) and belief systems become increasingly important. The analysis of spatial, material and social networks and their dynamics (Mills, 2017) aligns with other anthropological approaches to questions about the biocultural evolution of our species.

Ethnographic field studies are a major means for collecting data within cultural anthropological research. Although the method of participant-observation has well-known limitations (Howell, 2018), this is not the only tool in the anthropological tool kit. Complementing participant-observation are long-term and open-ended participatory methods that allow for individuals within the studied population to speak for themselves. Alongside these are tools like qualitative and structured interviews, genealogy and network analyses (Beer & König, 2020; Seligmann & Estes, 2020). Ethnology, the comparative study of peoples, represents another approach to answering canonical CES-inspired questions under the field study umbrella. Although ethnology has also long been subject to critique, it has never been absent from anthropological research (Schnegg, 2014). More recently, comparative methods have become supercharged by the increased accumulation and availability of cross-cultural data together with rigorous comparative methods (Fox & Gingrich, 2002). Such methods are now applied to a broad anthropological domain, expanded by inclusion of urban settings, the digital world and specific groups (e.g. professional, religious, gender) within larger communities. With a growing interest in the universality of human rights (Nagengast, 1997), relations of performance and cognition (Antweiler, 2016), and the origins and evolution of broader concepts such as religion and morality (Purzycki et al., 2018), more universal perspectives have been taken.

Contributions of evolutionary anthropology (EA) and its sister, human behavioural ecology (HBE) (differences between which are left aside here) prize coherence with other methods of knowledge

discovery and prioritize theoretical relations to natural selection. Unlike some areas of evolutionary science, work in EA and HBE generally avoids optimality assumptions that often characterize evolutionary psychology (about this see Fessler et al., 2016: 1030). The frameworks of EA and HBE occupy an unusually high-traffic crossroads within CES and, to extend the analogy, these crossroads are located near wide bridges traversing the waters between cultural and biological evolution. Despite their pluralistic use of methods and their ecumenical approach to research questions, EA and HBE are unified by the assumption, put roughly, that concepts from evolutionary biology have considerable utility in explanations of gene-culture co-evolutionary phenomena. Evolutionary anthropologists might study modifications of the hip, pelvis and spine revealed in the hominin archaeological record (Lovejoy, 2005a, b), or information about ancestral pathogen loads across regions in relation to present-day, culturally variable social psychological traits (Fincher et al., 2008; Faulkner et al., 2004). Modifications of the hominin body plan supporting bipedality, and cross-cultural variation in social-psychological factors in relation to pathogen load, are hypothesized to result, in the end, from their consequences on biological fitness. The ready explanatory pluralism of HBE and EA has probably contributed to their nuanced awareness of evolutionary trade-offs at play across human life history, e.g. between current and future reproduction and quantity and quality of offspring (Hill & Kaplan, 1999); their attention to both uniformity and difference between primate 'protoshame' and human shame systems (Fessler, 2007); and their pleasing, simultaneous concern with multiple dimensions of a single scientific explanation, dimensions such as physical realization, causation and phylogeny, which are often treated autonomously elsewhere in CES.

Working with results of field studies in the context of findings from elsewhere in CES makes a special contribution to challenges about knowledge synthesis and interdisciplinary integration (C5). This is not least because it is in the form of field studies that the bulk of research on non-human culture occurs. Operationalizing the culture concept across archaeology, biological anthropology and ethology, and applying it to behaviourally modern humans, earlier humans and non-humans, each requires improvements to this common but coarse-grained concept. Without more unanimity on the culture concept, survey results suggest that 'siloing' will continue (on the problems of disciplinary 'siloing' see Brewer et al., 2017: 2). Perhaps, for example, a law-like systematicity links the roles that social learning plays in the transmission of behavioural traits within different animal species and within hominin groups, but the underlying theory and methods issues have yet to be adequately resolved to support law-like cross-species generalizations about necessary conditions for the ontogeny of culture.

5. Unresolved theoretical issues in specific areas of cultural evolutionary science

In this section we focus on theoretical challenges faced in two core areas of CES: culture in other animals, and languages as culturally evolving systems. Our goal is just to show, with specific examples, that many unanswered or poorly framed scientific questions in cultural evolution would benefit from attention to theory.

5.1. Culture in other animals

The existence of behavioural traditions passing (not necessarily replicating) between generations through social learning in animals – a working characterization of 'culture' – is now uncontroversial. As a result, questions of interest to researchers have moved to focus on the mechanisms that support cultural transmission in animals, and questions about the methods with which animal cultures are studied. Answers to these questions can tell us whether human forms of culture – not least cumulative culture – are likely to be found in the animal kingdom.

Historically, animal cultures have been studied via the 'method of exclusion' (Whiten et al., 1999; later refined by Langergraber et al., 2011). This identifies behavioural patterns as cultural when they differ between geographically diverse groups of the same species. If genetic differences between these

groups can be ruled out (e.g. because of inter-group migration), and behavioural differences cannot be attributed to local ecological conditions, then behavioural patterns are likely to be products of social learning – and so are indicative of variant cultural traditions. While influential, this method has been criticized as insufficiently informative, and as resting on a conceptual problem, specifically a false dichotomy between nature and culture (Rogoff, 2003). First, it probably undercounts culture, where behaviours are universal (and so do not differ between populations) but are nonetheless socially learned. Second, it gives no insight into the kinds of social learning mechanisms that support cultural transmission (Sterelny, 2009; Acerbi et al., 2022). This criticism is important because some higher fidelity social learning processes, such as imitation (aka 'process copying' Buskell & Tennie, 2022, or 'know-how copying' Tennie, 2023), are thought to play an important (if not always necessary) role in the emergence of cumulative culture (Tennie et al., 2009; Moore, 2013). These may be missing in other species, or too inconsequential to produce and maintain similar cultures.

Tennie and colleagues have developed a series of paradigms designed to test whether putatively cultural behaviours in our nearest relatives, the great apes, are a product of imitation. Early paradigms established that chimpanzees do not acquire behaviours whose transmission requires high-fidelity transmission chains (Tennie et al., 2009). More recent studies show that many great ape behaviours can be explained as products of serial reinvention (often through trial-and-error learning) (e.g. Bandini et al., 2021). Since naïve apes spontaneously reinvent these behaviours when left alone with the necessary apparatus, they cannot depend on social learning. Tennie and colleagues call these behaviours 'latent solutions'. They do not deny that some forms of social learning support the transmission of great ape behaviours (Tennie et al., 2020). However, they argue that great apes do not copy new forms of 'know-how' from peers. Rather, for the most part, great ape social learning occurs because features of the environment are made salient through the actions of others. On this account, the presence of others in a learning environment is conceived of as a 'probabilistic difference maker' (Scarantino, 2015): a non-essential but nonetheless causally significant factor. It may facilitate the acquisition of behaviours without being essential for their development.

While it should now be uncontroversial that great apes do not, in their natural habitat, engage in high-fidelity know-how copying, the finding that great apes can learn to imitate (e.g. Pope et al., 2017) raises the question of whether their *lack* of imitation is itself a product of their developmental environment, and so potentially a culturally evolved 'cognitive gadget' (Heyes, 2018). Nonetheless, even trained apes seem unable to retain cultural inventions over successive generations (Tennie, 2019). This means that chimpanzees must often reinvent their ancestors' technologies in ways that human communities need not, and explains why cumulative culture is rare, at best, in great ape communities. Tentative reports of cumulative culture in whales and monkeys suggest that further empirical research is needed to determine whether and how animal cultures can gain in complexity over time. Theoretical research specifying what would constitute evidence of cumulative culture (and how 'acultural' explanations from natural selection would be ruled out) must precede or at least accompany additional field studies. A valuable contribution by theoreticians here would be to assist in fixing the social and cognitive conditions that are necessary and sufficient for cumulative culture, and by developing hypotheses about the kinds of mechanistic changes that support the development of higher fidelity social learning in non-human animals (Berio & Moore, 2023).

A final question asks whether non-human animals, and especially apes, are capable of understanding norms. Norm cognition is taken to be a prerequisite of many of our culturally evolved social institutions (e.g. institutions of justice). While some are sceptical that norm cognition is present in great apes (Schlingloff & Moore, 2017), others have argued that this scepticism is driven by an intellectualized account of the cognition required for normativity, suggesting a need for minimal models of norm cognition (Westra & Andrews, 2022). A further obstacle remains for investigating norm cognition in animals. The best method for identifying norms is through third party enforcement of perceived norm violations – and this is strikingly absent from great apes (Schlingloff & Moore, 2017). The theoretical formulation of alternative methods for identifying normative behaviours would thus constitute a valuable contribution to the field.

5.2. Languages as culturally evolving systems

Until not long ago, discussion of language 'evolution' was somewhat taboo. As recent as 1994, text-books observed that 'Evolution ... has become a "dirty word" in modern linguistic theory' (McMahon, 1994: 314). This is because through the twentieth century the nomenclature of evolution risked some allusion to outdated, often racist, views about how some languages – and by extension some social groups – might be 'more evolved' than others. It is now uncontroversial within linguistics that no language is inherently 'better' or 'more useful' than others. So, while vigilance must be maintained, the socio-political risks associated with evolutionary talk have begun to fade, and in the past 30 or so years evolutionary approaches to languages have grown dramatically in prominence (Tallerman & Gibson, 2011; Dediu & de Boer, 2016).

Today, the expression 'language evolution' is widely used to describe at least two distinct phenomena and one methodological advance. The two phenomena are: the biological process by which humans, and apparently only humans, became a 'language-ready' species; and the set of cultural – or 'cultural evolutionary' – processes by which relatively simple and unstructured systems become highly structured, and hence acquire some of the common, characteristic properties of languages, such as symbolism, compositionality and duality of patterning. The methodological advance is the application of phylogenetic tools, derived from population genetics, to study language history and language change. All three of these literatures are now very large, comprising thousands of papers each. Multiple past papers review, summarize and synthesize them (e.g. Mace & Holden, 2005; Tomasello & Call, 2019; Greenhill et al., 2020; Haspelmath, 2020; Roberts & Sneller, 2020; Scott-Phillips & Heintz, 2023; Scott-Phillips & Kirby, 2010). Here we focus on a philosophical issue arising from this growing influence of evolutionary thinking.

The challenge, put simply, is: what exactly does the evolutionary perspective bring that other approaches do not? The biology of language, language emergence and language change have all been important topics for language science for a long time, and a great deal of what has been uncovered by evolutionary approaches has previously been investigated and described in other terms. Research adopting an evolutionary perspective has enriched our understanding, provided new methods and added many new findings, but does it fundamentally alter our understanding of what the empirical phenomenon is and how it works?

One answer might be that a relatively faithful transposition of the Darwinian model – where variation, selection and inheritance combine to generate natural selection and hence the appearance of design in nature – is possible, and brings with it new insights, explanations and tools. With languages, there is something like variation (linguistic items vary enormously), there is something like selection (some items become more common than others) and there is something like inheritance (we learn from the previous generation). Some linguists indeed propose that linguistic items, of some sort or another, could be identified as units of selection closely analogous to genes (e.g. Croft, 2008; Ritt, 2004; Tamariz, 2019). However, deeper analysis raises difficult questions, and the issue is contentious: a decade ago, a group paper on language as a culturally evolving system observed that, 'Various scholars have proposed that concepts, cultural behaviors, or artifacts may function as replicators. It remains to be seen whether any, all, or some combination of these entities are reasonable candidates for cultural replicators' (Dediu et al., 2013: 314). The situation has not changed substantially since.

Another response to this question would be to adopt only specific parts of the Darwinian toolkit. That is, one could reduce the role of an analogy between variation in language and variation in genes while retaining other parts of Darwinism (Claidière et al., 2014). For instance, there might be scope for perspectives that adopt population thinking and some commitment to selectionism, but without the stronger commitment to replicators.

Languages are arguably the cultural domain par excellence for population thinking because languages are, more than any other cultural phenomena, especially amenable to 'atomization' – to being studied in terms of discrete items – and hence to identification and analysis of the types and their changing frequencies over time. As such, an emergent and pressing theoretical question is:

how best can linguistic types and tokens be precisely identified to facilitate cultural evolutionary study and modelling? Constructionist approaches to grammar, which describe grammars as networks of form–function pairs (Goldberg, 2003; Hilpert, 2014, *inter alia*), may be especially suitable for this goal. However, because there is little research developing this idea, treating constructions as the targets of population thinking (partial exceptions are Kirby, 1999; Steels, 2016) remains necessary.

6. Implications and applications of CET

The authors of the Cultural Evolution Society's report of survey results stated that one grand challenge involves the identification and study of cultural evolutionary processes that address 'significant social, economic, and political problems' (Brewer et al. 2017: 3), challenge (C8), which explicitly mentions climate change as an example. A growing number of scientists have begun focusing on such practical implications of CES, often concerning the role of norms. We define 'norms' as the shared values and rules that are enforced within a group, and which we contrast with the individual values or preferences of members. Reward and punishment are the basic means by which political entities and groups exert control over their members, organize them into roles and maintain order. For instance, reliable enforcement of norms against harming others reduces mutual fear among group members, making it easier to live together and making it more likely that individuals will survive. Indeed, from the perspective of both genetic and cultural evolution, theorists converge on cooperation and coordination as the general function of norms. Thus, practical implications of CES on social, economic and political challenges often concern the importance of norms in social life, yet coarse-grained concepts like norm or cooperation norm fall into the general conceptual problem mentioned above. Although modellers and experimentalists often refer to 'norms', the precise conditions under which norms (let alone different types of norms) are psychologically instantiated have yet to be perspicaciously individuated.

Adopting one approach to human cooperation defended within CES (Boyd & Richerson, 2009), philosophers Sripada and Stich (2006) attempted to address this challenge with the development of a theory of norm psychology that distinguishes between intrinsic and instrumental psychological motivations to follow and enforce norms. Instrumental motives focus on gaining rewards and avoiding punishment, while intrinsic motives concern the rightness or wrongness of an act itself, independently of any punishment or reward. A desire to comply with the speed limit may be instrumentally motivated, as a desire to avoid being fined as punishment, or it may be intrinsically motivated, as a desire to avoid doing something wrong by putting others at risk. Both motives can occur simultaneously, but research suggests that these are distinct functional capacities, with distinct inputs and outputs, and distinct ontogenies. Intrinsic motivations to follow norms evolved specifically in response to the adaptive problems posed by norm enforcement (Chudek & Henrich, 2011), so they could only have evolved quite recently, presumably after cultural learning capacities became sophisticated enough to support the widespread sharing of enforcement practices. This is neither the history nor the function of the various instrumental motivations involved in norm compliance, such as desire for social status, fear of pain, concern for mating prospects, the role of self-interest, etc. With a formal analysis of norms in hand, CES researchers could better test implications hypothesized to follow from one or another type of norm.

Davis et al. (2018) opened up this research by studying the intrinsic-instrumental motivational distinction as it operates within the sustainability movement. They discovered that too little attention is paid to the potential of intrinsic motives, and suggested clear ways to test the strength and frequency of intrinsic motives, and subsequently, methods to increase intrinsic motivation. Similarly, Kline et al. (2018) use the same theoretical framework to examine group structure and organization by analysing the norms of groups, organizations and institutions involved in suppressing free riding in sustainability contexts. For this they introduce novel models of cultural multilevel selection that address the competing pressures operating at different levels or scales of social organization. These attempts to apply CET to sustainability concerns have met with some uptake among practitioners in civil engineering, who face problems of free riding addressed at multiple levels of government. Hydrologists modelling

social-ecological systems have urged the adoption of a norm psychology framework informed by CES findings to address cooperation problems surrounding water scarcity and conservation (Yu et al., 2020).

The use of CET in sustainability science represents little more than proof of concept for the potential value of applying results in CES, and it is important not to dress old solutions up in the clothes of new theories (Chellappoo, 2022b). However, the case also illustrates a broader strategy of focusing theoretically perspicacious CES research upon norm-based social ills. For instance, as norms of democracy have come under threat from authoritarianism around the world (Norris & Inglehart, 2019), it has become salient that commitments to the rightness of democracy itself really matter. Without them, voters and officials often use their democratic powers to make governments less democratic or non-democratic. Addressing this problem requires the internalization of democratic norms, not instrumental reasons for following them. Another important form of intervention focuses on cases where conflicting norms exist in an equilibrium within a population. In such contexts changing the minds of a relatively small portion of the population can serve as a tipping point, initiating a cascade that quickly shifts the rest of the population to one side of the conflict (Andreoni et al., 2021). Against this background, Vogt et al. (2016) elegantly show how entertainment programming can be effective in shifting a few people's attitudes in one direction. In Sudan, they carefully embedded arguments against female genital cutting within a controlled series of telenovelas and found that viewers' attitudes toward uncut girls substantially improved. The demonstrably effective use of CES research to exploit mechanisms of cultural transmission to alter norms, and through them behaviour, augurs its own problems and might be received too optimistically (Wilson, 2011). This represents just one of several areas of CES in which knotty moral questions are rapidly arising, in this case about the role of CES in generating findings that allow political, corporate or military leaders to more effectively seed swiftly spreading, behaviour-altering cultural content through social and electronic media.

In addition to practical implications on individual attitudes and behaviours, cultural evolutionary processes have indelibly influenced human nature, for example by making human nature especially 'open-ended' (see Section 3 above), a continuing avenue for future CES and CET research (Hannon & Lewens, 2018; Ramsey, 2023). According to Kern and Moll (2017), the evolution of capacities for shared intentionality has transformed our species. Humans are the only species whose members have an awareness of themselves as embodying a unique, collectively structured form of life. The fact that culture accumulates over historical time has exerted remarkable effects on humans' understanding of self and other. There appears to be a sharp division between species whose culture 'ratchets up' in the way human culture does, and others, whose behavioural patterns are not preserved across generations. Importantly, with the dawn of cultural evolutionary processes, humans developed a consciousness of themselves as being part of a community whose members are interdependent in their learning (Moll & Kern, 2020; Moll, 2020). Bjorklund (2021) and other evo-devo scholars argue that what most drastically changed over the course of human evolution is our ontogeny, i.e. the processes by which we become mature individuals. We claim that hominin ontogeny was transformed such that children began to take a vested interest in their social learning and started to turn to adults as models and teachers. Adults, likewise, developed awareness of their role as teachers and authorities. Csibra and Gergely (2009, 2011) have discussed the evolution of a unique pedagogical communication system, whereby adults signal to learners their intent to transmit knowledge to them, and, in response, learners adopt a learning stance in which they assume the general relevance of what they are being shown or taught. Parents and their offspring thus not only became interdependent in unique ways but also developed an awareness of their interdependence and of their need to cooperate within a transmission system that serves to preserve lifeways that would otherwise disappear. This points to a possible constraint on cumulative cultural evolution worthy of further reflection: for culture to get preserved cross-generationally, there had to be a transformation in the cognitive apparatus and in the self-representation of the agents who are at once the 'movers' and 'recipients' of this process.

7. Conclusion

Although our review is broad, the field of cultural evolution is broader still, and several key areas and pressing problems were not adequately discussed above. We conclude by emphasizing two problems lurking near the ungainly relationship between cultural and biological evolution voiced in challenge (C2).

Since genes are fundamental units of selection in biological evolution, improving the relation between cultural and biological evolution, and integrating results across fields of inquiry inclusive of biology (see (C4) above), may necessitate increasing use of genetics research in CES. Scientists in cultural genomics research have identified countless neuro-functional differences across descent groups relevant for emotion and cognition at the levels of individuals and groups (Han, 2017; Chen & Moyzis, 2018). A recent, thorough review of genetic contributions to 'moral performance' found allelic variation in polymorphic genes linked to culturally characterizable descent groups. Functions of these polymorphisms related to neurohormone support, prosocial behaviour and aggressive behaviour. These neurohormones included oxytocin (regulated via polymorphic genes OXTR and CD38), dopamine (via COMPT, DRD3 and DRD4), serotonin (via TPH2 and 5-HTTLPR) and testosterone (via AR) (Wang & Su, 2022). Many moral cautions and epistemological objections accompany the use of gene × culture data, as they should. This area has met with its share of methodological missteps. For example, Chiao and Blizinsky's (2010) remarkable finding of a strong statistical relation between the ratio of several distinct serotonin transporter polymorphisms distributed across descent groups with the position of specific groups on the individualism-collectivism spectrum led authors to explain the relationship as a case of culturally infused biological selection. This result was subsequently invalidated (Eisenberg & Hayes, 2011). Even so, many findings in this field appear well warranted. Indeed, the statistical relationship just mentioned was later largely vindicated by Chiao, Blizinsky and co-authors through retesting on a much larger sample inclusive of not only European and East Asian but also African samples (Mrazek et al., 2013). We surmise that growth in this subfield is hindered by uncertainty and possible moral hazards. Compared with fields like anthropology and zoology, whose leading academic organizations have published statements governing research ethics, as a nascent field, cultural evolution may be relatively unprepared to deal with moral challenges implicitly raised by gene × culture research. At least, problems of this kind did not register as concerns in the Cultural Evolution Society's survey results (Brewer et al., 2017).

We finally turn to multi-level selection. The Cultural Evolution Society survey report identifies the topic of 'theory' and 'integration' as two of the most pressing issues facing the field (Brewer et al., 2017: 2). This pair forcefully collides when it comes to the status of multi-level selection theory in CES and evolutionary psychology. Recent work in evolutionary psychology contends that individual-level punishment explains substantial amounts of human cooperation (Krasnow et al., 2015); that third-party punishment is shown to arise from defence of personal interests (Kraswow et al., 2016); and that the threat of targeted conspiratorial killing selected for groupishness (Wrangham, 2021). Unifying these results is their support of the conclusion that biological selection on individual human organisms leads to group-level norms as by-products. These findings raise sceptical questions about the status of multi-level, especially group-level, adaptations in explanations of cooperation. Group-functional traits like language and norms are considered mere by-products of individual-level factors in evolutionary psychology (Gangestad et al., 2006, underappreciated in CES despite its authoritative authorship). In CES, though, group-level 'traits' are often treated as products of cultural adaptation and explanatorily irreducible to individual-level biological adaptations (Richerson et al., 2016). Evidence for group-level adaptations has been provided by theory (Sober & Wilson, 2003), modelling (Richerson et al., 2016) and experiment (Shariff & Norenzayan, 2007), but it is contested.

The disputed status of group selection and problems consequent from the use of this and allied coarse-grained terms is a significant cause of the wide gap and lack of knowledge integration between CES and fields like theoretical biology and evolutionary psychology and biology. This is epitomized by Krasnow and Delton's (2016: e43) argument that CES lacks a theoretical core capable of generating unambiguous, falsifiable predictions, is overly dependent on modelling and offers merely retrodictive

redescriptions of group-level traits as group adaptations. What developments could bring about a rapprochement between CES and the evolutionary human sciences? The study of evolutionary transitions in individuality holds prospects for more productive disagreement and engagement (see Carmel et al., 2023 and the issue of which it is a part). A more promising approach is likely to be creatively drawing and judiciously testing entailments of inclusive fitness theory to understand culture's role in cooperation – and biology's role in culture. (Of the 18,121 publications currently recorded in Web of Science as representing the 'topic' of 'cultural evolution', a mere 47 refer to 'inclusive fitness', suggesting untapped potential.) We suspect that, if the influence and authority of evolutionary anthropology and human behavioural ecology, pluralistic fields capable of a wide embrace, continue to grow within CES and CET, productive steps towards greater theoretical coherence with evolutionary sciences will follow.

Acknowledgements. The authors thank the John Templeton Foundation for funding the 'Philosophy of Science for Cultural Evolution' virtual research network (62377; Nichols, PI), which supported the writing of this article. We thank Andrew Buskell, Tim Lewens, Grant Ramsey, Kim Sterelny, two anonymous referees and the editor of this journal for valuable commentary on previous versions. The first author thanks Justin Barrett, Alex Arnold, Bryon Cunningham, and William Melanson for insightful conversations that improved thinking represented in this paper.

Author contributions. RN acquired funding to support the article and led its composition. All authors participated in the conception of and writing of sections of the article.

Financial support. Funding was received from John Templeton Foundation grant 62377.

Conflict of interest. The authors declare none.

Research transparency and reproducibility.. n/a.

References

Abrams, D. M., & Strogatz, S. H. (2003). Modelling the dynamics of language death. *Nature*, 424(6951), 900–900. https://doi.org/10.1038/424900a

Acerbi, A., Ghirlanda, S., & Enquist, M. (2014). Regulatory traits: Cultural influences on cultural evolution. In S. Cagnoni, M. Mirolli, & M. Villani (Eds.), *Evolution, complexity and artificial life* (pp. 135–147). Springer. https://doi.org/10.1007/978-3-642-37577-4_9

Acerbi, A., Snyder, W. D., & Tennie, C. (2022). The method of exclusion (still) cannot identify specific mechanisms of cultural inheritance. *Scientific Reports*, 12(1), 21680.

Akbari, M., Bahrami-Rad, D., & Kimbrough, E. O. (2019). Kinship, fractionalization and corruption. *Journal of Economic Behavior & Organization*, 166, 493–528. https://doi.org/10.1016/j.jebo.2019.08.012

Andreoni, J., Nikiforakis, N., & Siegenthaler, S. (2021). Predicting social tipping and norm change in controlled experiments. *Proceedings of the National Academy of Sciences*, 118(16), e2014893118.

Antweiler, C. (2016). Local knowledge as a universal social product: A general model and a case from Southeast Asia. In P. Meusburger, T. Freytag, & L. Suarsana (Eds.), Ethnic and cultural dimensions of knowledge (pp. 165–190). Springer.

Axelrod, R., & Hamilton, W. D. (1981). The evolution of cooperation. Science, 211(4489), 1390–1396. https://doi.org/10.1126/science.7466396

Bandini, E., Grossmann, J., Funk, M., Albiach-Serrano, A., & Tennie, C. (2021). Naïve orangutans (Pongo abelii and Pongo pygmaeus) individually acquire nut-cracking using hammer tool. American Journal of Primatology, 83(9), e23304.

Bartlett, F. C. (1932). Remembering: A study in experimental and social psychology. Cambridge University Press.

Beer, B., & König, A. (Eds.). (2020). Methoden ethnologischer Feldforschung. Reimer.

Bentley, R. A., Hahn, M. W., & Shennan, S. J. (2004). Random drift and culture change. Proceedings of the Royal Society of London. Series B: Biological Sciences, 271(1547), 1443–1450.

Berio, L., & Moore, R. (2023). Great ape enculturation studies: A neglected resource in cognitive development research. Biology & Philosophy, 38(2), 17.

Bjorklund, D. (2021). How children invented humanity. The role of development in human evolution. Oxford University Press. Blythe, R. A. (2012). Neutral evolution: A null model for language dynamics. Advances in Complex Systems, 15(03n04), 1150015

Borg, J. M., Buskell, A., Kapitany, R., Powers, S. T., Reindl, E., & Tennie, C. (2022). Evolved open-endedness in cultural evolution: A new dimension in open-ended evolution research. ArXiv Preprint ArXiv:2203.13050.

Boyd, R., & Richerson, P. J. (1985). Culture and the evolutionary process. University of Chicago Press.

- Boyd, R., & Richerson, P. J. (1996). Why culture is common, but cultural evolution is rare. *Proceedings of the British Academy*, 88, 77–93.
- Boyd, R., & Richerson, P. J. (2009). Culture and the evolution of human cooperation. Philosophical Transactions of the Royal Society B: Biological Sciences, 364(1533), 3281–3288. https://doi.org/10.1098/rstb.2009.0134
- Boyer, P., & Ramble, C. (2001). Cognitive templates for religious concepts: Cross-cultural evidence for recall of counter-intuitive representations. Cognitive Science, 25(4), 535–564. https://doi.org/10.1207/s15516709cog2504_2
- Brewer, J., Gelfand, M., Jackson, J. C., MacDonald, I. F., Peregrine, P. N., Richerson, P. J., ..., Wilson, D. S. (2017). Grand challenges for the study of cultural evolution. *Nature Ecology & Evolution*, 1(3), 0070. https://doi.org/10.1038/s41559-017-0070
- Broesch, J., Barrett, H. C., & Henrich, J. (2014). Adaptive content biases in learning about animals across the life course. Human Nature, 25, 181–199. https://doi.org/10.1007/s12110-014-9191-1
- Brown, C. (2007). Differential equations: A modeling approach (Vol. 150). Sage.
- Buskell, A. (2017). What are cultural attractors? Biology and Philosophy, 32(3), 377–394. https://doi.org/10.1007/s10539-017-9570-6
- Buskell, A. (2019). Looking for middle ground in cultural attraction theory. Evolutionary Anthropology: Issues, News, and Reviews, 28(1), 14–17. https://doi.org/10.1002/evan.21762
- Buskell, A. (2022). Cumulative culture and complex cultural traditions. Mind & Language, 37(3), 284-303.
- Buskell, A., Enquist, M., & Jansson, F. (2019). A systems approach to cultural evolution. *Palgrave Communications*, 5(1), 1–15. Buskell, A., & Tennie, C. (2022). Mere recurrence and cumulative culture at the margins. *The British Journal for the Philosophy of Science*, 717776. https://doi.org/10.1086/717776
- Caldwell, C. A., & Millen, A. E. (2008). Experimental models for testing hypotheses about cumulative cultural evolution. *Evolution and Human Behavior*, 29(3), 165–171. https://doi.org/10.1016/j.evolhumbehav.2007.11.003
- Carmel, Y., Shavit, A., Lamm, E., & Szathmáry, E. (2023). Human socio-cultural evolution in light of evolutionary transitions: Introduction to the theme issue. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 378(1872), 20210397. https://doi.org/10.1098/rstb.2021.0397
- Cavalli-Sforza, L. L., & Feldman, M. W. (1973). Cultural versus biological inheritance: Phenotypic transmission from parents to children. (A theory of the effect of parental phenotypes on children's phenotypes). American Journal of Human Genetics, 25(6), 618–637.
- Cavalli-Sforza, L. L., & Feldman, M. W. (1981). Cultural transmission and evolution: A quantitative approach. Princeton University Press.
- Centola, D., Gonzalez-Avella, J. C., Eguiluz, V. M., & San Miguel, M. (2007). Homophily, cultural drift, and the co-evolution of cultural groups. *Journal of Conflict Resolution*, 51(6), 905–929.
- Charbonneau, M. (2015). All innovations are equal, but some more than others: (Re)integrating modification processes to the origins of cumulative culture. *Biological Theory*, 10(4), 322–335.
- Charbonneau, M. (2016). Modularity and recombination in technological evolution. *Philosophy & Technology*, 29(3), 373–392. Charbonneau, M. (2020). Understanding cultural fidelity. *British Journal for the Philosophy of Science*, 71(4), 1209–1233.
- Charbonneau, M., & Bourrat, P. (2021). Fidelity and the grain problem in cultural evolution. Synthese, 199, 5815–5836.
- Chellappoo, A. (2021). Rethinking prestige bias. Synthese, 198(9), 8191-8212.
- Chellappoo, A. (2022a). Cultural evolution: A case study in global epistemologies of science. In D. Ludwig, I. Koskinen, Z. Mncube, L. Poliseli, & L. Reyes-Galindo (Eds.), Global epistemologies and philosophies of science (pp. 208–219). Routledge. Chellappoo, A. (2022b). When can cultural selection explain adaptation? Biology & Philosophy, 37(1), 2.
- Chen, C.-Y., & Moyzis, R. K. (2018). Cultural genomics: Promises and challenges. *Journal of Cross-Cultural Psychology*, 49(5), 764–788.
- Chiao, J. Y., & Blizinsky, K. D. (2010). Culture–gene coevolution of individualism–collectivism and the serotonin transporter gene. Proceedings of the Royal Society B: Biological Sciences, 277(1681), 529–537. https://doi.org/10.1098/rspb.2009.1650
- Chowdhury, L. S., Zhang, Y., & Nichols, R. (2022). Footbinding and its cessation: An agent-based model adjudication of the labor market and evolutionary sciences hypotheses. *Evolution and Human Behavior*, 43(6), 475–489. https://doi.org/10.1016/j.evolhumbehav.2022.08.005
- Chudek, M., & Henrich, J. (2011). Culture-gene coevolution, norm-psychology and the emergence of human prosociality. Trends in Cognitive Sciences 15(5), 218–226.
- Claidière, N., Scott-Phillips, T. C., & Sperber, D. (2014). How Darwinian is cultural evolution? Philosophical Transactions of the Royal Society B, 369, 20130368.
- Creanza, N., Kolodny, O., & Feldman, M. W. (2017). Cultural evolutionary theory: How culture evolves and why it matters. Proceedings of the National Academy of Sciences, 114(30), 7782–7789. https://doi.org/10.1073/pnas.1620732114
- Croft, W. (2008). Evolutionary linguistics. Annual Review of Anthropology, 37, 219-234.
- Cronk, L. (2007). The influence of cultural framing on play in the trust game: A Maasai example. *Evolution and Human Behavior*, 28(5), 352–358.
- Csibra, G., & Gergely, G. (2009). Natural pedagogy. Trends in Cognitive Sciences, 13(4), 148–153. https://doi.org/10.1016/j.tics. 2009.01.005

- Csibra, G., & Gergely, G. (2011). Natural pedagogy as evolutionary adaptation. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 366(1567), 1149–1157. https://doi.org/10.1098/rstb.2010.0319
- Dapschauskas, R., Göden, M. B., Sommer, C., & Kandel, A. W. (2022). The emergence of habitual ochre use in Africa and its significance for the development of ritual behavior during the Middle Stone Age. *Journal of World Prehistory*, 35(2), 233–319.
- Davis, T., Hennes, E. P., & Raymond, L. (2018). Cultural evolution of normative motivations for sustainable behaviour. Nature Sustainability, 1(5), 218–224. https://doi.org/10.1038/s41893-018-0061-9
- Dawkins, R. (1989). The selfish gene (New ed). Oxford University Press.
- Dediu, D., Cysouw, M., Levinson, S. C., Baronchelli, A., Christiansen, M. H., Croft, W., ... Lieven, E. (2013). Cultural evolution of language. In *Cultural evolution: Society, technology, language & religion*. Strüngmann Forum Reports (Vol. 12, pp. 303–332). MIT Press.
- Dediu, D., & de Boer, B. (2016). Language evolution needs its own journal. Journal of Language Evolution, 1(1), 1-5.
- Dennett, D. C. (2017). From bacteria to Bach and back: The evolution of minds. WW Norton.
- Denton, K. K., Ram, Y., Liberman, U., & Feldman, M. W. (2020). Cultural evolution of conformity and anticonformity. Proceedings of the National Academy of Sciences, 117(24), 13603–13614. https://doi.org/10.1073/pnas.2004102117
- Derex, M., Beugin, M. P., Godelle, B., & Raymond, M. (2013). Experimental evidence for the influence of group size on cultural complexity. *Nature*, 503(7476), 389–391.
- Derex, M., Godelle, B., & Raymond, M. (2014). How does competition affect the transmission of information? *Evolution and Human Behavior*, 35(2), 89–95. https://doi.org/10.1016/j.evolhumbehav.2013.11.001
- Driscoll, C. (2011). Fatal attraction? Why Sperber's attractors do not prevent cumulative cultural evolution. *British Journal for the Philosophy of Science*, 62(2), 301–322.
- Driscoll, C. (2017). The evolutionary culture concepts. *Philosophy of Science*, 84(1), 35–55. https://doi.org/10.1086/688934 Efferson, C., Lalive, R., Richerson, P., Mcelreath, R., & Lubell, M. (2008). Conformists and mavericks: The empirics of frequency-dependent cultural transmission★. *Evolution and Human Behavior*, 29(1), 56–64. https://doi.org/10.1016/j. evolhumbehav.2007.08.003
- Eisenberg, D. T. A., & Hayes, M. G. (2011). Testing the null hypothesis: Comments on Culture–gene coevolution of individualism–collectivism and the serotonin transporter gene. *Proceedings of the Royal Society B: Biological Sciences*, 278(1704), 329–332. https://doi.org/10.1098/rspb.2010.0714
- Faulkner, J., Schaller, M., Park, J. H., & Duncan, L. A. (2004). Evolved disease-avoidance mechanisms and contemporary xenophobic attitudes. *Group Processes & Intergroup Relations*, 7(4), 333–353. https://doi.org/10.1177/1368430204046142
- Feldman, M. W., & Cavalli-Sforza, L. L. (1976). Cultural and biological evolutionary processes, selection for a trait under complex transmission. Theoretical Population Biology, 9(2), 238–259. https://doi.org/10.1016/0040-5809(76)90047-2
- Fessler, D. (2007). From appeasement to conformity: Evolutionary and cultural perspectives on shame, competition, and cooperation. In J. L. Tracy, R. W. Robins, & J. P. Tangney (Eds.), *The self-conscious emotions: Theory and research* (pp. 174–193). Guilford Press.
- Fessler, D. M. T., Clark, J. A., & Clint, E. K. (2016). Evolutionary psychology and evolutionary anthropology. In D. M. Buss (Ed.), *The handbook of evolutionary psychology* (2nd ed., Vol. 2, pp. 1029–1046). John Wiley & Sons.
- Fincher, C. L., Thornhill, R., Murray, D. R., & Schaller, M. (2008). Pathogen prevalence predicts human cross-cultural variability in individualism/collectivism. *Proceedings of the Royal Society B: Biological Sciences*, 275(1640), 1279–1285. https://doi.org/10.1098/rspb.2008.0094
- Fishwick, P. A. (1995). Simulation model design and execution: Building digital worlds. Prentice Hall.
- Fox, R. G., & Gingrich, A. (Eds.). (2002). Anthropology, by comparison. Routledge.
- Fracchia, J., & Lewontin, R. C. (1999). Does culture evolve? History and Theory, 38(4), 52-78.
- Galen, L. W. (2016). Big gods: Extended prosociality or group binding? Behavioral and Brain Sciences, 39. https://doi.org/10. 1017/S0140525X15000436
- Gangestad, S., Haselton, M., & Buss, D. (2006). Evolutionary foundations of cultural variation: Evoked culture and mate preferences. Psychological Inquiry, 17(2), 75–95.
- Gärdenfors, P., & Högberg, A. (2017). The archaeology of teaching and the evolution of *Homo docens*. Current Anthropology, 58(2), 188–208.
- Gerkey, D. (2013). Cooperation in context: Public goods games and post-Soviet collectives in Kamchatka, Russia. *Current Anthropology*, 54(2), 144–176.
- Godfrey-Smith, P. (2001). On the status and explanatory structure of developmental systems theory. In S. Oyama, R. D. Gray, & P. E. Griffiths (Eds.), Cycles of contingency: Developmental systems and evolution (pp. 283–297). MIT Press.
- Godfrey-Smith, P. (2012). Darwinism and cultural change. Philosophical Transactions of the Royal Society B, 367(1599), 2160–2170.
- Goldberg, A. E. (2003). Constructions: A new theoretical approach to language. Trends in Cognitive Sciences, 7(5), 219–224.
 Gomes, C. M., & McCullough, M. E. (2015). The effects of implicit religious primes on dictator game allocations: A preregistered replication experiment. Journal of Experimental Psychology: General, 144(6), e94–e104. https://doi.org/10.1037/xge0000027
- Gray, R.D. and Watts, J. (2017). Cultural macroevolution matters. Proceedings of the National Academy of Sciences, 114(30), 7846–7852.

- Greenhill, S. J., Heggarty, P., & Gray, R. D. (2020). Bayesian phylolinguistics. *The handbook of historical linguistics*, 2, 226–253. Griesemer, J. R. (1988). Genes, memes, and demes. *Biology and Philosophy*, 3, 179–184.
- Hagen, E. H., & Hammerstein, P. (2006). Game theory and human evolution: A critique of some recent interpretations of experimental games. Theoretical Population Biology, 69(3), 339–348.
- Han, S. (2017). The sociocultural brain: A cultural neuroscience approach to human nature. Oxford University Press.
- Hannon, E., & Lewens, T. (2018). Why we disagree about human nature (1st ed). Oxford University Press.
- Haspelmath, M. (2020). Human linguisticality and the building blocks of languages. Frontiers in Psychology, 10. doi:10.3389/fpsyg.2019.03056
- Henrich, J. (2001). Cultural transmission and the diffusion of innovations: Adoption dynamics indicate that biased cultural transmission is the predominate force in behavioral change. *American Anthropologist*, 103(4), 992–1013. https://doi.org/10.1525/aa.2001.103.4.992
- Henrich, J. (2004). Demography and cultural evolution: How adaptive cultural processes can produce maladaptive losses the Tasmanian case. *American Antiquity*, 69(02), 197–214. https://doi.org/10.2307/4128416
- Henrich, J., Boyd, R., Derex, M., Kline, M. A., Mesoudi, A., Muthukrishna, M., ..., Thomas, M. G. (2016). Understanding cumulative cultural evolution. *Proceedings of the National Academy of Sciences*, 113(44). https://doi.org/10.1073/pnas.1610005113 Henrich, J., Boyd, R., & Richerson, P. J. (2008). Five misunderstandings about cultural evolution. *Human Nature*, 19(2), 119–
- Henrich, J., Boyd, R., & Richerson, P. J. (2008). Five misunderstandings about cultural evolution. *Human Nature*, 19(2), 119-137. https://doi.org/10.1007/s12110-008-9037-1
- Henrich, J., & McElreath, R. (2007). Dual-inheritance theory: The evolution of human cultural capacities and cultural evolution. In L. Barrett & R. Dunbar (Eds.), Oxford handbook of evolutionary psychology (pp. 555–570). Oxford University Press.
- Heyes, C. M. (2018). *Cognitive gadgets: The cultural evolution of thinking*. The Belknap Press of Harvard University Press. Hill, K., & Kaplan, H. (1999). Life history traits in humans: Theory and empirical studies. *Annual Review of Anthropology*, 28 (1), 397–430. https://doi.org/10.1146/annurev.anthro.28.1.397
- Hilpert, M. (2014). Construction grammar & its application to English. Edinburgh University Press.
- Howell, S. (2018). Ethnography. In F. Stein (Ed.), The open encyclopedia of anthropology. Facsimile of the first edition in The Cambridge encyclopedia of anthropology. https://doi.org/10.29164/18ethno
- Hull, D. L. (1988). A mechanism and its metaphysics: An evolutionary account of the social and conceptual development of science. Biology and Philosophy, 3, 123–155.
- Hunt, G. R., & Gray, R. D. (2003). Diversification and cumulative evolution in New Caledonian crow tool manufacture. *Proceedings of the Royal Society of London. Series B: Biological Sciences*, 270(1517), 867–874.
- Jiménez, Á. V., & Mesoudi, A. (2019). Prestige-biased social learning: Current evidence and outstanding questions. *Palgrave Communications*, 5(1), 1–12.
- Jiménez, Á. V., & Mesoudi, A. (2020). Prestige does not affect the cultural transmission of novel controversial arguments in an online transmission chain experiment. *Journal of Cognition and Culture*, 20(3–4), 238–261.
- Johnson, D. D. P. (2005). God's punishment and public goods: A test of the supernatural punishment hypothesis in 186 world cultures. *Human Nature*, 16(4), 410–446. https://doi.org/10.1007/s12110-005-1017-0
- Kandler, A., & Laland, K. N. (2009). An investigation of the relationship between innovation and cultural diversity. *Theoretical Population Biology*, 76(1), 59–67.
- Kandler, A., Perreault, C., & Steele, J. (2012). Cultural evolution in spatially structured populations: A review of alternative modeling frameworks. Advances in Complex Systems, 15(01n02), 1203001.
- Kelemen, D. (1999). Why are rocks pointy? Children's preference for teleological explanations of the natural world. Developmental Psychology, 35(6), 1440–1452. https://doi.org/10.1037/0012-1649.35.6.1440
- Kern, A., & Moll, H. (2017). On the transformative character of collective intentionality and the uniqueness of the human. *Philosophical Psychology*, 30(3), 315–333. https://doi.org/10.1080/09515089.2017.1295648
- Killin, A., & Pain, R. (2023). How WEIRD is Cognitive Archaeology? Engaging with the Challenge of Cultural Variation and Sample Diversity. *Review of Philosophy and Psychology*, 14(2), 539–563. https://doi.org/10.1007/s13164-021-00611-z
- Kirby, S. (1999). Function, selection, and innateness: The emergence of language universals. Oxford University Press.
- Kirby, S., Griffiths, T., & Smith, K. (2014). Iterated learning and the evolution of language. *Current Opinion in Neurobiology*, 28, 108–114.
- Kline, M. A., Waring, T. M., & Salerno, J. (2018). Designing cultural multilevel selection research for sustainability science. Sustainability Science, 13(1), 9–19.
- Krasnow, M. M., & Delton, A. W. (2016). The sketch is blank: No evidence for an explanatory role for cultural group selection. Behavioral and Brain Sciences, e43, 39.
- Krasnow, M. M., Delton, A. W., Cosmides, L., & Tooby, J. (2015). Group cooperation without group selection: Modest punishment can recruit much cooperation. *PLOS ONE*, 10(4), e0124561.
- Krasnow, M. M., Delton, A. W., Cosmides, L., & Tooby, J. (2016). Looking under the hood of third-party punishment reveals design for personal benefit. *Psychological Science*, 27(3), 405–418. https://doi.org/10.1177/0956797615624469
- Lang, M., Purzycki, B. G., Apicella, C. L., Atkinson, Q. D., Bolyanatz, A., Cohen, E., ..., Henrich, J. (2019). Moralizing gods, impartiality and religious parochialism across 15 societies. *Proceedings of the Royal Society B: Biological Sciences*, 286 (1898), 20190202. https://doi.org/10.1098/rspb.2019.0202

- Langergraber, K. E., Boesch, C., Inoue, E., Inoue-Murayama, M., Mitani, J. C., Nishida, T., ..., Vigilant, L. (2011). Genetic and 'cultural' similarity in wild chimpanzees. *Proceedings of the Royal Society B: Biological Sciences*, 278(1704), 408–416. https://doi.org/10.1098/rspb.2010.1112
- Legare, C. H. (2017). Cumulative cultural learning: Development and diversity. Proceedings of the National Academy of Sciences, 114(30), 7877–7883. https://doi.org/10.1073/pnas.1620743114
- Lewens, T. (2015). Cultural evolution: Conceptual challenges. Oxford University Press.
- Lewens, T. (2020). How can conceptual analysis contribute to scientific practice?: The case of cultural evolution. In K. Kampourakis & T. Uller (Eds.), *Philosophy of science for biologists* (1st ed., pp. 146–167). Cambridge University Press. https://doi.org/10.1017/9781108648981.009
- Lewens, T., & Buskell, A. (2023). Cultural evolution. In E. Zalta & U. Nodelman (Eds.), Stanford encyclopedia of philosophy. https://plato.stanford.edu/archives/sum2023/entries/evolution-cultural/
- Lightner, A. D., Barclay, P., & Hagen, E. H. (2017). Radical framing effects in the ultimatum game: The impact of explicit culturally transmitted frames on economic decision-making. *Royal Society Open Science*, 4(12), 170543.
- Lightner, A. D., Bendixen, T., & Purzycki, B. G. (2023). Moralistic supernatural punishment is probably not associated with social complexity. *Evolution and Human Behavior*, 44(6), 555–565.
- Lombard, M., & Haidle, M. N. (2012). Thinking a bow-and-arrow set: Cognitive implications of Middle Stone Age bow and stone-tipped arrow technology. Cambridge Archaeological Journal, 22(2), 237–264.
- Lovejoy, C. O. (2005a). The natural history of human gait and posture. Gait & Posture, 21(1), 113–124. https://doi.org/10.1016/j.gaitpost.2004.06.010
- Lovejoy, C. O. (2005b). The natural history of human gait and posture. *Gait & Posture*, 21(1), 95–112. https://doi.org/10.1016/j.gaitpost.2004.01.001
- Mace, R., & Holden, C. J. (2005). A phylogenetic approach to cultural evolution. *Trends in Ecology & Evolution*, 20(3), 116–121.
- Manrubia, S. C., & Zanette, D. H. (2002). At the boundary between biological and cultural evolution: The origin of surname distributions. *Journal of Theoretical Biology*, 216(4), 461–477.
- McElreath, R., & Henrich, J. (2007). Modelling cultural evolution. In L. Barrett & R. Dunbar (Eds.), Oxford handbook of evolutionary psychology (pp. 555–570). Oxford University Press.
- McGrew, W. C. (1992). Chimpanzee material culture: Implications for human evolution. Cambridge University Press.
- McMahon, A. M. S. (1994). Understanding language change. Cambridge University Press.
- Mesoudi, A. (2011). Cultural evolution. University of Chicago Press.
- Mesoudi, A. (2016). Cultural evolution: A review of theory, findings and controversies. *Evolutionary Biology*, 43(4), 481–497. https://doi.org/10.1007/s11692-015-9320-0
- Mesoudi, A., Chang, L., Murray, K., & Lu, H. J. (2014). Higher frequency of social learning in China than in the West shows cultural variation in the dynamics of cultural evolution. *Proceedings of the Royal Society B: Biological Sciences*, 282(1798), 20142209–20142209. https://doi.org/10.1098/rspb.2014.2209
- Mesoudi, A., & Whiten, A. (2008). The multiple roles of cultural transmission experiments in understanding human cultural evolution. Philosophical Transactions of the Royal Society B: Biological Sciences, 363(1509), 3489–3501.
- Micheletti, A. J. C., Brandl, E., & Mace, R. (2022). What is cultural evolution anyway? *Behavioral Ecology*, 33(4), 667–669. https://doi.org/10.1093/beheco/arac011
- Mills, B. J. (2017). Social network analysis in archaeology. Annual Review of Anthropology, 46, 379-397.
- Misra, A. K. (2012). A simple mathematical model for the spread of two political parties. *Nonlinear Analysis: Modelling and Control*, 17(3), 343–354.
- Miton, H., & Charbonneau, M. (2018). Cumulative culture in the laboratory: Methodological and theoretical challenges. Proceedings of the Royal Society B: Biological Sciences, 285(1879), 20180677. https://doi.org/10.1098/rspb.2018.0677
- Miyatake, S., & Higuchi, M. (2017). Does religious priming increase the prosocial behaviour of a Japanese sample in an anonymous economic game?: Religion and prosociality in Japan. *Asian Journal of Social Psychology*, 20(1), 54–59. https://doi.org/10.1111/ajsp.12164
- Moll, H. (2020). How young children learn from others. Journal of Philosophy of Education, 54(2), 340-355.
- Moll, H., & Kern, A. (2020). Learning from another. Inquiry. https://doi.org/10.1080/0020174X.2020.1731593
- Moore, R. (2013). Imitation and conventional communication. Biology & Philosophy, 28, 481-500.
- Mörchen, J., Luhn, F., Wassmer, O., Kunz, J. A., Kulik, L., Van Noordwijk, ..., Schuppli, C. (2023). Migrant orangutan males use social learning to adapt to new habitat after dispersal. Frontiers in Ecology and Evolution, 11, 1158887. https://doi.org/ 10.3389/fevo.2023.1158887
- Mrazek, A. J., Chiao, J. Y., Blizinsky, K. D., Lun, J., & Gelfand, M. J. (2013). The role of culture-gene coevolution in morality judgment: Examining the interplay between tightness-looseness and allelic variation of the serotonin transporter gene. *Culture and Brain*, 1(2–4), 100–117. https://doi.org/10.1007/s40167-013-0009-x
- Mullon, C., & Lehmann, L. (2017). Invasion fitness for gene-culture co-evolution in family-structured populations and an application to cumulative culture under vertical transmission. Theoretical Population Biology, 116, 33-46. https://doi. org/10.1016/j.tpb.2017.06.003

- Nagengast, C. (1997). Women, minorities, and indigenous peoples: Universalism and cultural relativity. *Journal of Anthropological Research*, 53(3), 349–369.
- Nichols, R., Slingerland, E., Nielbo, K. L., Kirby, P., & Logan, C. (2020). Supernatural agents and prosociality in historical China: Micro-modeling the cultural evolution of gods and morality in textual corpora. *Religion, Brain & Behavior*. https://doi.org/10.1080/2153599X.2020.1742778
- Norenzayan, A. (2013). Big gods: How religion transformed cooperation and conflict. Princeton University Press.
- Norenzayan, A., Atran, S., Faulkner, J., & Schaller, M. (2006). Memory and mystery: The cultural selection of minimally counterintuitive narratives. *Cognitive Science*, 30(3), 531–553. https://doi.org/10.1207/s15516709cog0000_68
- Norris, P., & Inglehart, R. (2019). Cultural backlash: Trump, Brexit, and authoritarian populism. Cambridge University Press. O'Brien, M. J., Buchanan, B., & Eren, M. I. (2018). Convergent evolution in stone-tool technology. MIT Press.
- Pain, R., and R. L. Brown (2021). Mind the gap: A more evolutionarily plausible role for technical reasoning in cumulative technological culture. In F. Stein (Ed.), *The open encyclopedia of anthropology*. Facsimile of the first edition in *The Cambridge encyclopedia of anthropology*, 199(1–2), 2467–2489.
- Van Dyke Parunak, H., Savit, R., & Riolo, R. L. (1998). Agent-based modeling vs. equation-based modeling: A case study and users' guide. Multi-Agent Systems and Agent-Based Simulation: First International Workshop, MABS'98, Paris, France, July 4–6, 1998. Proceedings (Vol. 1, pp. 10–25).
- Patten, A. (2014). Equal recognition: The moral foundations of minority rights. Princeton University Press. https://doi.org/10. 1515/9781400850433.
- Perrault, C. (2019). The quality of the archaeological record. University of Chicago Press.
- Pisor, A. C., Gervais, M. M., Purzycki, B. G., & Ross, C. T. (2020). Preferences and constraints: The value of economic games for studying human behaviour. *Royal Society Open Science*, 7(6), 192090.
- Pope, S. M., Taglialatela, J. P., & Hopkins, W. D. (2017). Changes in mirror region connectivity following do-as-i-do training in chimpanzees. *Journal of Cognitive Neuroscience*, 30(3), 1–11.
- Purzycki, B. G., Henrich, J., Apicella, C. L., Atkinson, Q. D., Baimel, A., Cohen, E., & Norenzayan, A. (2018). The evolution of religion and morality: A synthesis of ethnographic and experimental evidence from eight societies. *Religion, Brain & Behavior*, 8(2), 101–132.
- Purzycki, B. G., Stagnaro, M. N., & Sasaki, J. (2020). Breaches of trust change the content and structure of religious appeals. Journal for the Study of Religion, Nature and Culture, 14(1), 71–94.
- Purzycki, B. G., & Willard, A. K. (2016). MCI theory: A critical discussion. *Religion, Brain & Behavior*, 6(3), 207–248. https://doi.org/10.1080/2153599X.2015.1024915
- Ramsey, G. (2023). Human nature. Cambridge University Press.
- Ramsey, G., Bastian, M. L., & Van Schaik, C. P. (2007). Animal innovation defined and operationalized. Behavioral and Brain Sciences, 30, 393–437.
- Ramsey, G., & De Block, A. (2015). Is cultural fitness hopelessly confused? The British Journal for the Philosophy of Science, axv047. https://doi.org/10.1093/bjps/axv047
- Read, D. (2011, January 1). The misuse of a mathematical model: The Tasmanian case. Human Complex Systems UCLA, 2011. https://ssrn.com/abstract=2805822 or http://dx.doi.org/10.2139/ssrn.2805822
- Reali, F., & Griffiths, T. L. (2009). Words as alleles: Connecting language evolution with Bayesian learners to models of genetic drift. *Proceedings of the Royal Society B: Biological Sciences*, 277(1680), 429–436.
- Richerson, P., Baldini, R., Bell, A. V., Demps, K., Frost, K., Hillis, V., ... Zefferman, M. (2016). Cultural group selection plays an essential role in explaining human cooperation: A sketch of the evidence. *Behavioral and Brain Sciences*, 39, e30. https://doi.org/10.1017/S0140525X1400106X
- Ritt, N. (2004). Selfish sounds & linguistic evolution: A Darwinian approach to language change. Cambridge University Press.
- Roberts, G., & Sneller, B. (2020). Empirical foundations for an integrated study of language evolution. Language Dynamics and Change, 10(2), 188–229.
- Roes, F., & Raymond, M. (2003). Belief in moralizing gods. Evolution and Human Behavior, 24(2), 126–135. https://doi.org/10.1016/S1090-5138(02)00134-4
- Rogoff, B. (2003). The cultural nature of human development. Oxford University Press.
- Rowthorn, R. (2011). Religion, fertility and genes: A dual inheritance model. Proceedings of the Royal Society B: Biological Sciences, 278(1717), 2519–2527.
- Scarantino, A. (2015). Information as a probabilistic difference maker. Australasian Journal of Philosophy, 93(3), 419-443.
- Schelling, T. C. (1971). Dynamic models of segregation. Journal of Mathematical Sociology, 1(2), 143-186.
- Schlingloff, L., & Moore, R. (2017). Do chimpanzees conform to social norms? In *The Routledge handbook of philosophy of animal minds* (pp. 381–389). Routledge.
- Schloss, J. P., & Murray, M. J. (2011). Evolutionary accounts of belief in supernatural punishment: A critical review. *Religion, Brain & Behavior*, 1(1), 46–99. https://doi.org/10.1080/2153599X.2011.558707
- Schnegg, M. (2014). Anthropology and comparison: Methodological challenges and tentative solutions. Zeitschrift Für Ethnologie, 55–72.

- Schulze, J. N., Denkinger, J., Oña, J., Poole, M. M., & Garland, E. C. (2022). Humpback whale song revolutions continue to spread from the central into the eastern South Pacific. Royal Society Open Science, 9(8), 220158.
- Schuppli, C., & Van Schaik, C. P. (2019). Animal cultures: How we've only seen the tip of the iceberg. *Evolutionary Human Sciences*, 1, e2.
- Scott-Phillips, T., & Heintz, C. (2023). Great ape interaction: Ladyginian but not Gricean. *Proceedings of the National Academy of Sciences*, 120(42), e2300243120.
- Scott-Phillips, T. C., & Kirby, S. (2010). Language evolution in the laboratory. Trends in Cognitive Sciences, 14(9), 411–417. https://doi.org/10.1016/j.tics.2010.06.006
- Segovia-Martin, J., & Tamariz, M. (2021). Synchronising institutions and value systems: A model of opinion dynamics mediated by proportional representation. *PLoS One*, 16(9), e0257525.
- Segovia-Martín, J., Walker, B., Fay, N., & Tamariz, M. (2020). Network connectivity dynamics, cognitive biases, and the evolution of cultural diversity in round-robin interactive micro-societies. *Cognitive Science*, 44(7), e12852.
- Seligmann, L. J., & Estes, B. P. (2020). Innovations in ethnographic methods. American Behavioral Scientist, 64(2), 176–197.
 Shariff, A., Norenzayan, A., & Henrich, J. (2010). The birth of high gods: How the cultural evolution of supernatural policing influenced the emergence of complex, cooperative human societies, paving the way for civilization. In M. Schaller, A. Norenzayan, S. J. Heine, T. Yamagishi, & T. Kamedda (Eds.), Evolution, culture, and the human mind (pp. 119–136). Psychology Press.
- Shariff, A. F., & Norenzayan, A. (2007). God is watching you: Priming god concepts increases prosocial behavior in an anonymous economic game. *Psychological Science*, 18(9), 803–809. https://doi.org/10.1111/j.1467-9280.2007.01983.x
- Shennan, S. (2008). Evolution in archaeology. Annual Review of Anthropology, 37, 75-91.
- Singh, M., Acerbi, A., Caldwell, C. A., Danchin, É., Isabel, G., Molleman, L., ... Derex, M. (2021). Beyond social learning. Philosophical Transactions of the Royal Society B: Biological Sciences, 376(1828), rstb.2020.0050, 20200050. https://doi. org/10.1098/rstb.2020.0050
- Skyrms, B. (2014). Evolution of the social contract. Cambridge University Press.
- Slingerland, E., Atkinson, Q. D., Ember, C. R., Sheehan, O., Muthukrishna, M., Bulbulia, J., & Gray, R. D. (2020). Coding culture: Challenges and recommendations for comparative cultural databases. *Evolutionary Human Sciences*, 2, e29. https://doi.org/10.1017/ehs.2020.30
- Smith, D. (2020). Cultural group selection and human cooperation: A conceptual and empirical review. *Evolutionary Human Sciences*, 2, e2. https://doi.org/10.1017/ehs.2020.2
- Smolla, M., Jansson, F., Lehmann, L., Houkes, W., Weissing, F. J., Hammerstein, P., ... Enquist, M. (2021). Underappreciated features of cultural evolution. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 376(1828), rstb.2020.0259, 20200259. https://doi.org/10.1098/rstb.2020.0259
- Sober, E., & Wilson, D. S. (2003). Unto others: The evolution and psychology of unselfish behavior (4th ed.). Harvard University Press.
- Sperber, D. (1996). Explaining culture: A naturalistic approach. Blackwell.
- Sperber, D. (2000). An objection to the memetic approach to culture. In R. Aunger (Ed.), Darwinizing culture: The status of memetics as a science (pp. 163–173). Oxford University Press.
- Sripada, C. S., & Stich, S. (2006). A framework for the psychology of norms. In *The innate mind Vol. 2: Cognition and culture* (pp. 280–301). Oxford University Press.
- Steels, L. (2016). Agent-based models for the emergence and evolution of grammar. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 371(1701), 20150447.
- Sterelny, K. (2009). Peacekeeping in the culture wars. In K. N. Laland & B. G. Galef (Eds.), *The question of animal culture* (pp. 288–304). Harvard University Press.
- Sterelny, K. (2012). The evolved apprentice. MIT Press.
- Sterelny, K. (2017). Cultural evolution in California and Paris. Studies in History and Philosophy of Biological and Biomedical Sciences, 62, 42–50.
- Sterelny, K. (2021). The Pleistocene social contract. Oxford University Press.
- Stolarczyk, R. E., & Schmidt, P. (2018). Is early silcrete heat treatment a new behavioural proxy in the Middle Stone Age? *PLoS One*, 13(10), e0204705.
- Swanson, G. E. (1960). The birth of the gods: The origin of primitive beliefs. Univ. of Michigan Press.
- Tallerman, M., & Gibson, K. R. (Eds.). (2011). The Oxford Handbook of Language Evolution. Oxford University Press.
- Tamariz, M. (2017). Experimental studies on the cultural evolution of language. Annual Review of Linguistics, 3, 389–407.
- Tamariz, M. (2019). Replication and emergence in cultural transmission. Physics of Life Reviews, 30, 47-71.
- Tennie, C. (2019). Could non-human great apes also have cultural evolutionary psychology? *Behavioral and Brain Sciences*, 42, e184.
- Tennie, C. (2023). Reinforcing a Zone of Latent Solution (ZLS) account of the Oldowan. A commentary on Sterelny and Hiscock's 'Cumulative culture, archaeology, and the Zone of Latent Solutions'. *Current Anthropology*, forthcoming.
- Tennie, C., Bandini, E., Van Schaik, C. P., & Hopper, L. M. (2020). The zone of latent solutions and its relevance to understanding ape cultures. *Biology & Philosophy*, 35(1), 1–42.

- Tennie, C., Call, J., & Tomasello, M. (2009). Ratcheting up the ratchet: On the evolution of cumulative culture. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 364(1528), 2405–2415.
- Tennie, C., Premo, L. S., Braun, D. R., & McPherron, S. P. (2017). Early stone tools and cultural transmission: Resetting the null hypothesis. *Current Anthropology*, 58(5), 652–672. https://doi.org/10.1086/693846
- Tomasello, M., & Call, J. (2019). Thirty years of great ape gestures. Animal cognition, 22, 461-469.
- Tunçgenç, B., & Cohen, E. (2016). Movement synchrony forges social bonds across group divides. Frontiers in Psychology, 7, 782.Turchin, P., Whitehouse, H., Francois, P., Slingerland, E., & Collard, M. (2012). A historical database of sociocultural evolution. Cliodynamics, 3(2).
- Vaesen, K., Collard, M., Cosgrove, R., & Roebroeks, W. (2016a). Population size does not explain past changes in cultural complexity. Proceedings of the National Academy of Sciences, 113(16). https://doi.org/10.1073/pnas.1520288113
- Vaesen, K., Collard, M., Cosgrove, R., & Roebroeks, W. (2016b). Reply to Henrich et al.: The Tasmanian effect and other red herrings. Proceedings of the National Academy of Sciences, 113(44), E6726–E6727. https://doi.org/10.1073/pnas.1613074113
 Vaesen, K., & Houkes, W. (2021). Is human culture cumulative? Current Anthropology, 62(2), 218–238.
- Vale, G. L., Flynn, E. G., Kendal, R., Rawlings, B., Hopper, L. M., Schapiro, S. J., ..., Kendal, R. L. (2017). Testing differential use of payoff-biased social learning strategies in children and chimpanzees. *Proceedings of the Royal Society B: Biological Sciences*, 284(1868), 20171751.
- Vanhaeren, M., & d'Errico, F. (2006). Aurignacian ethno-linguistic geography of Europe revealed by personal ornaments. Journal of Archaeological Science, 33(8), 1105–1128.
- Vogt, S., Mohmmed Zaid, N. A., El Fadil Ahmed, H., Fehr, E., & Efferson, C. (2016). Changing cultural attitudes towards female genital cutting. *Nature*, 538(7626), 506–509.
- Wang, Y., & Su, Y. (2022). Genetic contributions to East Asian morality. In R. Nichols (Ed.), *The Routledge international handbook of morality, cognition, emotion, and behavior in China* (pp. 80–100). Routledge.
- Waring, T. M., Goff, S. H., & Smaldino, P. E. (2017). The coevolution of economic institutions and sustainable consumption via cultural group selection. *Ecological Economics*, 131, 524–532.
- Watts, J., Greenhill, S. J., Atkinson, Q. D., Currie, T. E., Bulbulia, J., & Gray, R. D. (2015). Broad supernatural punishment but not moralizing high gods precede the evolution of political complexity in Austronesia. *Proceedings of the Royal Society B: Biological Sciences*, 282(1804), 20142556–20142556. https://doi.org/10.1098/rspb.2014.2556
- Weisberg, M. (2007). Who is a modeler? The British Journal for the Philosophy of Science, 58(2), 207-233.
- Westra, E., & Andrews, K. (2022). A pluralistic framework for the psychology of norms. *Biology & Philosophy, 37*(5), 40. Whiten, A. (2019). Cultural evolution in animals. *Annual Review of Ecology, Evolution, and Systematics, 50*(1), 27–48. https://doi.org/10.1146/annurev-ecolsys-110218-025040
- Whiten, A. (2021). The burgeoning reach of animal culture. Science, 372(6537), eabe6514.
- Whiten, A., Goodall, J., McGrew, W. C., Nishida, T., Reynolds, V., Sugiyama, Y., ..., Boesch, C. (1999). Cultures in chimpanzees. *Nature*, 399(6737), 682–685. https://doi.org/10.1038/21415
- Wilson, D. S. (2011). The neighborhood project: Using evolution to improve my city, one block at a time (1st ed.). Little, Brown. Wimsatt, W. C. (1999). Genes, memes, and cultural heredity. Biology & Philosophy, 14(2), 279–310. https://doi.org/10.1023/A:1006646703557
- Wolf, E. R. (2005). Europe and the people without history: With a new preface (reprinted). University of California Press. Wood, C., Caldwell-Harris, C., & Stopa, L. (2018). The rhythms of discontent: Synchrony impedes performance and group functioning in an interdependent coordination task. Journal of Cognition and Culture, 18(1-2), 154-179.
- Wrangham, R. W. (2021). Targeted conspiratorial killing, human self-domestication and the evolution of groupishness. *Evolutionary Human Sciences*, 3, e26.
- Yu, D., Chang, H., Davis, T., Hillis, V., Marston, L., Oh, W. S., ... Waring, T. M. (2020). Socio-hydrology: An interplay of design and self-organization in a multilevel world. *Ecology and Society*, 25(4).